Do Robots Necessarily Displace Workers? Evidence from Mexican Local Labor Markets

Joaquín Serrano*

May 30, 2024

[Click here for most recent version]

Abstract

This study uses the rapid rise of automation in Mexico to examine the labor market effects of industrial robot adoption in industrializing countries. I combine unique data on robot imports with administrative, survey, and economic census data. I exploit spatial and time variation in robot imports across commuting zones to estimate the impacts of automation on labor market outcomes. I find that industrial robot adoption does not negatively affect formal employment and average wages in Mexico; these results contrast with findings for developed countries. I can rule out adverse effects of a size as large as -0.7% for formal employment and -0.01% for formal wages in the short run, and I find positive effects of 6% and 3%, respectively, in the long run. I also provide evidence that these results are not due to compositional effects. Additionally, I show that after a CZ starts importing robots, its aggregated gross value added per worker increases, consistent with the productivity effect dominating or compensating for any adverse displacement effect. Despite a substantial pool of unskilled labor, automation in industrializing economies such as Mexico can yield nonnegative employment effects because of the significant productivity enhancements associated with a higher price elasticity of demand and the early stages of automation.

Keywords: Technology, Robot Adoption, Local Labor Markets, Mexico. **JEL Classification**: J23, J24, J31, J46, O14, O17, R10.

^{*}Joaquín Serrano: Department of Economics, UCLA. Email: serranojqn@g.ucla.edu. Website: serranojqn.com. I am very grateful to Adriana Lleras-Muney, Daniel Haanwinckel, Natalie Bau, and Martha Bailey for their continual support and guidance throughout this research project. I thank David Jaume, Juliana Londoño-Vélez, Michela Giorcelli, and Dora Costa for their comments and helpful conversations. I thank Diana Flores Peregrina, Fatih Ozturk, Jonathan Kowarski, David Henning, Victoria Wang, Fernanda Rojas-Ampuero, Victoria Barone, Ariadna Jou, and participants in the Albert Family Fund Proseminar in Applied Microeconomics at UCLA for insightful discussions. I benefited from facilities and resources provided by the California Center for Population Research at UCLA (CCPR), which receives core support (P2C-HD041022, NICHD).

1 Introduction

Machines—from assembly lines and electric motors to computers and modern robots have fundamentally reshaped workplaces. In 2022, a staggering 3.5 million industrial robots were operational, with a remarkable annual growth rate in this number of 31% (IFR, 2022). As robotics, artificial intelligence, and computing power advance, there is a growing fear of significant job losses (Acemoglu & Restrepo, 2020; Arntz, Gregory, & Zierahn, 2019; Frey & Osborne, 2017). These concerns prompt essential questions about the societal implications of automation. However, amid the uncertainty, this technological shift also offers the potential to boost productivity. Previous research has predominantly examined industrialized countries, demonstrating that automation reduces employment and wages, particularly among low-skilled workers engaged in routine tasks (Acemoglu & Restrepo, 2020, 2022a; Chiacchio, Petropoulos, & Pichler, 2018; Dauth, Findeisen, Suedekum, & Woessner, 2021).

Theoretically, the expected sign of the net effect of automation on outcomes is ambiguous (Acemoglu & Restrepo, 2019, 2020). On the one hand, automation replaces workers (the displacement effect), leading to declines in employment and wages in the short run. On the other hand, automation reduces production costs and allows firms to lower prices and increase sales (the productivity effect), leading to increases in employment and wages. Productivity effects in industrializing countries may be larger because automation occurs in sectors exposed to a more competitive product market such that demand is more sensitive to price changes.¹ In addition, these economies are at an early stage of automation where there is more room for large marginal gains. However, empirical evidence on the effects of automation in these economies is still lacking because it is a relatively recent process and data on robot adoption are scarce.

This paper overcomes these challenges by providing empirical evidence on the labor market effects of automation in Mexico. Mexico serves as an ideal case study for several reasons. First, the country has witnessed a remarkable surge in industrial robot adoption over the past decade, with an average annual growth rate of 40% since 2011. This transformation is especially significant within the manufacturing sector, which accounts for a large share of employment.² Second, Mexico provides administrative data on industrial

¹The productivity effect will be larger, especially when product demand is relatively more elastic. In developing countries, robot adoption is more prevalent in export manufacturing, which is exposed to foreign competition, and therefore demand is arguably more elastic. Firms in this sector are more likely to invest in automation because of the relatively large fixed cost and the associated product quality upgrade, particularly when interdependence of components is a critical factor (Artuc, Bastos, Copestake, & Rijkers, 2022). Moreover, domestic consumers have relatively low incomes and are more likely to be price sensitive.

²Mexico is also among the most populous Latin American countries, second only to Brazil, with a pop-

robot imports, including the location and timing of robot adoption at fine geographic levels. These data measure robot adoption much more precisely in a given area than the samples in previous works, which typically rely on time series data by industry at the national level. I complement these with publicly available administrative data from the Mexican Social Security Institute (IMSS, for its Spanish acronym) to examine labor market outcomes in the formal sector, supplemented by insights from the National Occupation and Employment Survey (ENOE) on the informal sector and from Mexican economic censuses on changes in productivity and balance sheet information across all Mexican firms. These high-quality data allow a comprehensive examination of the effects of robot adoption at the local labor market level.

Identifying the causal effects of automation is challenging because of a combination of data limitations and endogeneity issues. Endogeneity concerns such as omitted variable and simultaneity biases complicate the analysis. Consider, for example, a scenario where an industry faces a negative labor supply shock and turns to automation in part to mitigate rising labor costs. Neglecting this dynamic would overestimate the negative effects of automation on employment. Conversely, if industries facing positive product-market shocks or relaxation of their financial constraints adopt robots to help them increase production, neglecting this dynamic would lead to overestimation of the positive impact of automation on employment.

To address the endogeneity issue, I employ multiple complementary strategies. My primary approach is to take advantage of timing-based event study methodologies. I identify an event as the first year in which a commuting zone (CZ) receives industrial robot imports, and then, I group CZs into cohorts based on these event years. To mitigate selection effects, I limit the sample to only CZs that import robots at some point. Therefore, the comparison group includes only CZs that import robots but do so later than the CZ under consideration. This method assumes that the timing of events is random and is not influenced by anticipation (pretreatment periods are not affected by future events) and that the outcomes for CZs importing robots and the comparison group follow parallel trends. I explore any systematic link between importation timing and observable factors to validate this assumption. Furthermore, I include granular industry-level trends to mitigate potential biases and account for industry-specific shocks, such as increasing competition. Assuming that the comparison groups and the specific fixed effects and trends account for unobservable shocks, I exploit the remaining exogenous variation arising from a combination of changes in robot supply (cost reduction and improved capa-

ulation of approximately 126 million people, 60 million of whom are of working age, among whom 62% participate in the labor force.

bilities) and exogenous variation in exposure across Mexican regions due to supply-side factors, like supply chain disruptions, random events at ports or shipping facilities, or differences in trade networks. For instance, when the cost of robots produced in Korea falls, producers with existing connections to Korean robot producers are likely to benefit first.

I find that the rapid adoption of robots in Mexico in the last decade did not lead to significant negative effects on formal employment or average real formal wages. The effect of robot adoption on employment is zero in the first year of adoption. Then, it stays stable for the first four years, with an average effect of 1.6% that is not statistically significant at the 5% level. I can rule out negative effects larger than -0.7% on formal employment and larger than -0.01% on formal wages for the average of these four years. I also provide evidence that these results do not hide compositional effects of the CZs' cohorts around the event time. This zero effect could hide potential heterogeneity across groups. However, there are, at most, minor negative effects, even among groups expected to be more exposed to the negative effects of robot adoption, based on findings from industrialized countries. These groups include workers in the manufacturing sector, workers in smaller CZs, medium-wage and prime-age workers, and workers in occupations related to industrial robots.

When I estimate the effects over an average of 5 to 8 years after the event, they consistently remain nonnegative, increase in magnitude, and achieve statistical significance. However, it is important to acknowledge that changes in composition might influence these long-term effects. This is because cohorts experience the event in different years and are observed over varying time frames after the event. Consequently, I interpret these results cautiously, concluding that automation may not lead to significant job displacement effects. This highlights that robot adoption does not necessarily harm workers and could hold transformative potential for labor market dynamics.

The main concern with my estimation method is that the adoption of robots is endogenous, and in particular could be related to positive demand shocks. To address this, I use alternative methods, such as event study approaches, triple differences (DDD), and synthetic difference-in-differences (SDD). The DDD method compares manufacturing industries that are high robot adopters with non-manufacturing non-robot adopters industries within each CZ. This allows me to net out specific shocks at the CZ level, such as positive growth trends for all the sectors in regions adopting robots first. This fact is important since, a priori, we might expect that growing CZs would adopt robots first. The SDD enables me to address potential violations of the identifying assumption of parallel trends. The method selects a different set of controls for each treated unit, composed of a weighted average of CZ that matches that unit's pre-trend. Therefore, the control group is essentially individualized, and I can investigate treatment effect heterogeneity across cohorts, relevant to analyze long-term effects avoiding compositional changes. I consistently observe the same qualitative patterns as the baseline estimates across different methods and outcome measures. In addition, I conduct a sensitivity analysis that considers varying degrees of potential violations of parallel trends in post-event periods. The results confirm the robustness of the main findings: a substantial deviation from parallel trends would be required to yield negative employment and wage estimates.

The results are also robust when employing different measures of robot adoption. One might suspect that the nonnegative effect is due to the need for a certain number of industrial robots in the CZ. Therefore, I define the event based on the cumulative value of robot imports surpassing a predefined threshold. Alternatively, I use a measure of intensity based on the cumulative number of years that a CZ receives imports. The analysis is also robust if I use alternative measures for my primary outcomes, formal employment, and average wages.

To explore potential mechanisms underlying industrial robot adoption, I employ Mexican economic censuses covering the universe of Mexican firms and balance sheet information. Within this analysis, I consistently observe that robot adoption enhances gross value added (GVA) per worker across all industries, consistent with productivity increases. The effect on the labor share is negative but imprecisely estimated. When I restrict the sample to the manufacturing sectors more likely to adopt automation, I find an increase in GVA per worker, input cost, payroll, and fixed assets, indicating increased productivity, wages, and investment. Despite these gains, the automation effects on the labor share remain negative but not statistically significant. In nonmanufacturing sectors, adopting robots generally results in statistically nonsignificant effects; however, when statistically significant, these effects tend to be small and lack a systematic pattern. These findings suggest that productivity effects dominate or compensate for potential adverse displacement effects in the case of Mexico in the short to medium run.

This study contributes to the literature on automation's impact on local labor markets, not only in developed countries (Acemoglu & Restrepo, 2020; Chen & Frey, 2021; Chiacchio et al., 2018; Dauth et al., 2021; Graetz & Michaels, 2018) but also in industrializing economies (Brambilla, César, Falcone, & Gasparini, 2023; Calì & Presidente, 2021a; Rodrigo, 2021), for which the evidence is still very limited and not conclusive.³ Most

³Brambilla et al. (2023) examines the impact of domestic robots on labor markets in Argentina, Brazil,

of these papers rely on industry-level proxies of robot adoption, which assume uniformity in robot adoption within a particular industry across different districts, as implied by Bartik measures, generating potential attenuation bias. In addition, this data do not allow for exploiting variation within industry in the adoption of robots –variation that I can exploit in my setting and which allows me to control for industry–year, CZ–industry, and CZ–year specific shocks in some specifications. Because of this more granular level of variation I can control for additional shocks and therefore obtain potentially different estimates. I complement this literature by leveraging administrative data with precise information on the timing and location of domestic automation adoption, which allows me not to rely on shift-share measures and to use alternative empirical strategies.

There are several studies analyzing the impact of US automation on developing countries through trade linkages and reshoring (Artuc et al., 2022; Kugler, Kugler, Ripani, & Rodrigo, 2021; Stemmler, 2020). Two evaluate automation in Mexico (Artuc, Christiaensen, & Winkler, 2019; Faber, 2020) during 1990–2015 and include exposure to domestic automation in their analysis, using similar industry-level data, Bartik instruments, and periods. However, they arrive at different conclusions about the wage-employment effects of the increase in domestic robots, ranging from null to negative effects. My paper complements these studies by using district-level data on robot adoption, introducing a novel empirical strategy, and focusing on a more recent period marked by accelerated domestic robot adoption in Mexico. It also offers a context where domestic production of robots is minimal and firms primarily depend on imports, a scenario different from the ones in previous studies centered on countries with active domestic robot production (Acemoglu et al., 2022; Acemoglu, Lelarge, & Restrepo, 2021; Dixon, Hong, & Wu, 2019; Humlum, 2020; Koch, Manuylov, & Smolka, 2021). These contributions enrich our understanding of how automation is shaping the future of work and economic development in Mexico and beyond, offering valuable insights for policymakers.

The paper is structured as follows. Section 2 describes the data sources, including industrial robot adoption and labor market outcomes, while offering descriptive evidence that sheds light on the association between these two factors. Section 3 gives an overview of the institutional background of the labor market and the rise of automation in Mexico. Section 4 describes the empirical strategy for identifying the impact of industrial robot adoption. Section 5 presents the main results on labor market outcomes, and Section 6

and Mexico, finding increases in unemployment and informality but no statistically significant negative effect on the employment rate and wages. Rodrigo (2021) identifies positive effects on firms' productivity but no aggregate employment impact in Brazil. For Indonesia, Ing and Zhang (2022) and Calì and Presidente (2021b) observe significant productivity and employment gains from automation in manufacturing.

shows results on local mechanisms that may explain the effects of industrial robots. Section 7 presents a conceptual framework and discusses why the effects of automation could differ in developing countries such as Mexico from those found for developed countries. Section 8 concludes.

2 Data

To study the effect of robot adoption on Mexican labor markets, I compile data from multiple sources: (i) detailed administrative records on the universe of industrial robot imports in Mexico provided by the Ministry of Economy; (ii) data on formal employment, publicly accessible and released by the Mexican Social Security Institute (*Instituto Mexicano del Seguro Social*, IMSS); (iii) the Mexican employment survey (ENOE); (iv) information from the 1990 and 2000 population censuses; and (v) information from economic censuses spanning from 2003 to 2018.

I conduct the analysis at the local labor market level.⁴ There are many ways of defining labor markets; generally, they are defined based on local administrative (state or municipality) boundaries, but these only sometimes coincide with economically relevant boundaries. Therefore, I use commuting zones (CZs) as the unit of observation, grouping 2438 municipalities (similar to US counties) into 1806 labor markets following Faber (2020). His algorithm follows a similar procedure designed by Atkin (2016), which clusters municipalities into CZs based on the intensity of commuting between them.⁵

⁴Identifying the relevant units of analysis is crucial for analyzing the effects of robot adoption on different outcomes. Firms that adopt robots can extend their impact to other related firms, within their industry or even within their labor market by sharing the same labor and product suppliers, consumers, and competitors (Acemoglu, Koster, & Ozgen, 2023; Acemoglu & Restrepo, 2022b; Aghion, Antonin, Bunel, & Jaravel, 2022). Making comparisons solely between robot-adopting and nonadopting firms within a single labor market could yield biased estimates of the impact of automation. To prevent this issue, one should conduct the analysis at the labor market level, capturing the direct effect of robot adopters and the indirect impact on other economic agents within the same labor market. Furthermore, supporting evidence indicates that workers do not exhibit perfect mobility across geographic regions in the US (Autor & Dorn, 2013); the case of Mexico is likely to be similar in this regard.

⁵The algorithm operates as follows: (i) clustering all municipalities within a metropolitan area into one larger municipality; (ii) computing the intensity of commuting from each municipality to any other municipality by dividing the number of people who commute by the number of residents in the origin municipality; and (iii) clustering municipalities into CZs if more than 10% of residents of either municipality commute into the other. See Faber (2020) for more details.

2.1 Robot adoption data

To measure robot adoption at the labor market level, I use information on the universe of industrial robot imports from customs data collected by the Mexican Ministry of Economy. I follow other papers that use a similar measure for developed countries (Acemoglu et al., 2023; Bessen, Goos, Salomons, & van den Berge, 2020) and define robots using the code 847950 from the international trade codes of commodities (Harmonized System, 2012), which is defined as "industrial robots, not elsewhere specified or included."⁶ According to the International Standards Organizations, an "industrial robot is an actuated mechanism programmable in two or more axes, with a degree of autonomy, moving within its environment, to perform intended tasks."

The data provide information about the month and municipality in which the robots were internationally purchased, the country of origin, and the import value. In this paper, I use the publicly available version of these data, which are aggregated to the municipality–month level. The aggregated import value is available if at least three firms imported in the given period. As explained previously, I map municipalities to CZs based on the intensity of commuting flows between them. This transition from municipality–month to CZ–year aggregation allows a more comprehensive examination of the effects of automation on local labor markets. The dataset covers the period from 2006 to the present.

My data on robot imports at the municipality level allow me to measure the exposure to robot adoption more accurately than has been done in previous literature. The standard approach in the literature is to use a Bartik-type measure, which relies on national industry variation in robot shipments (based on data from the International Federation of Robots (IFR)) and the industry composition of each commuting zone (Acemoglu & Restrepo, 2020; Dauth, S Findeisen, Suedekum, & Woessner, 2018; Graetz & Michaels, 2018). This approach assumes that the same number of robots per worker are installed in each industry across all commuting zones, which may not be realistic and could generate biased estimates. In addition, in the case of Mexico, the IFR data start in 2011 and are aggregated along with the US and Canada for previous periods. My data can overcome

⁶The Harmonized System (HS) is an internationally recognized product classification system that designates unique six-digit codes for various categories and commodities. HS code 847950 specifically pertains to industrial robots not categorized under any other code. These robots are capable of autonomously performing a range of tasks, such as welding, painting, assembling, or handling material, without the need for human intervention. Industrial robots falling under this HS code encompass a variety of machines, including robotic arms employed for manipulating objects or tools in manufacturing settings and robotic vehicles designed for transporting materials or goods within warehouses or production facilities. For more information, see WCO (2013), and visit the website https://www.wcotradetools.org.

these limitations, as they capture the actual location and year of robot imports over an extended period.

My measure of robot adoption indicates whether the CZ is receiving industrial robot imports in a given year. From the data, I observe 104 different CZs receiving at least one industrial robot importation during 2006–2022. I exclude from the sample the CZs receiving imports in 2006 since this is the first year with available data and I assume that these CZs were receiving robot imports even before 2006. Figure 2 displays the geographic distribution of robot imports across CZs within the restricted sample. The robot imports are distributed across the entire territory, with the Bajío area and the northern region containing 46% of the CZs that receive robot imports, followed by the southern central region (15%) and the northeastern region (13%).⁷ Within these regions, the percentage of CZs receiving imports varies from 3% to 13%, providing ample geographical variation for estimating effects.

Using custom records may have limitations; for example, it is possible that firms obtain robots from local manufacturers instead of importing them. However, this concern is insignificant in Mexico, where almost all robot adopters import them from the US, Japan, or Europe.⁸ To validate the accuracy of the customs data, I compare them with national-level IFR data, finding a solid 98% correlation between the number of importing municipalities and the stock of robot installations over the period 2011–2020 (see Figure 1 for a comparison of robot adoption measures).^{9,10} This distinctive feature of my research setting is advantageous because, in the developed countries studied in similar works, robot manufacturing is substantial and robot imports capture only a fraction of total robot adoptions.

Another potential concern with the use of import records is the possible concentration of robots in cities where firms' headquarters are located rather than where their plants

⁷The Bajío area includes Aguascalientes, Guanajuato, Querétaro, San Luis Potosí, and Zacatecas. The northern region comprises Baja California, Baja California Sur, Chihuahua, Durango, and Sinaloa. The southern central region encompasses Hidalgo, Puebla, Tlaxcala, and Veracruz de Ignacio de la Llave. The northeastern region includes Coahuila de Zaragoza, Nuevo León, and Tamaulipas.

⁸Global robot manufacturing is highly concentrated, with the top ten suppliers in 2021 being Japan, Germany, Italy, China, the United States, Denmark, France, South Korea, Austria, and Sweden, which accounted for 80% of total robot exports. In contrast, Mexico ranks 24th, contributing just 0.4% to global industrial robot exports, with a trade deficit in this sector.

⁹I obtain a similarly strong correlation if I compare the IFR stock of robots with the stock of the number of importing firms or the stock of the importation values.

¹⁰The IFR defines an industrial robot as automatically controlled (not manually operated through a joystick or pushbuttons), reprogrammable without physical alteration, multipurpose-manipulable, programmable in three or more axes, fixed in place or mobile, and for use in industrial automation applications. The definition does not include machines such as elevators, ATMs, smart washing machines, transportation tools, textile looms, software, or autonomous cars.

are situated. In Mexico, it is relatively uncommon for firms that use robots to have multiple plants. Among the firms that imported robots between 2006 and 2022, a significant majority, approximately 85%, have only one plant, and 88% operate within the same CZ. Additionally, since the automotive sector is responsible for most robot adoption in Mexico, I identify the universe of large automotive plants and verify that most are located in CZs receiving robot imports (see Figure 2).

To address the possibility that firms import machinery through domestic robot integrators specializing in installing robots at local plants, I examine the practices in Mexico to understand whether this is a significant issue. My investigation reveals that the leading robot integrators in Mexico do not import robots or any other type of machines.

2.2 Labor market outcomes

Formal labor market: To assess the impact of industrial robot adoption on formal employment, I leverage a publicly accessible administrative dataset released monthly by the IMSS. This dataset provides municipality-level information on the total count of registered (i.e., formal sector) workers and total payroll by firm size, industry (up to the 4-digit sector), gender and wage categories. The IMSS administers healthcare services to formal employees in the private sector, and because of the mandatory nature of contributions from private firms, it holds comprehensive information on their workers.

Using these data, I compute the total number of formal employees in a given CZ and 4-digit sector and the corresponding average wage by dividing the total payroll by the total number of workers. Although the frequency of the data is monthly, I focus on the information in December of each year. For the primary analysis, I do not restrict the sample by age. However, the data allow me to distinguish employee age groups and investigate potential heterogeneity.

One evident caveat of this dataset is that it does not include the public or informal sector, which is relatively large in Mexico. The informality rate in Mexico between 2005 and 2018 varied from 56% to 59% of the employed population, according to INEGI. Nevertheless, drawing from prior research, most firms that import and adopt industrial robots are more likely to be large in terms of both employees and revenues (Acemoglu et al., 2022; Koch et al., 2021) and therefore are less likely to be in the informal sector (Porta & Shleifer, 2008, 2014; Ulyssea, 2018). Another limitation of the dataset is its lack of information on hours worked, part-time or full-time employment, specific occupations or tasks, and other characteristics of workers such as educational attainment, ethnicity, and immigration status. However, the IMSS data allow me to dividing employees into different wage groups, in terms of the Mexican minimum wage level.¹¹

Mexican labor force survey (ENOE). To shed light on the effects of robot adoption on transitions between the formal and informal sectors, into unemployment, and in and out of the labor force, I use the Mexican labor force survey, called ENOE. This survey is at quarterly frequency and covers most of the localities in urban centers, which I aggregate into CZs. To maintain comparability with the IMSS data, I focus on the fourth quarter of each year between 2005 and 2020. I calculate the count of formal and informal employees, unemployed individuals, and workers out of the labor force. I define formal workers as those affiliated with IMSS and informal workers as those lacking affiliation with any social security subsystems (IMSS, PEMEX, or ISSSTE). However, given ENOE's different sample of localities, I repeat my primary analysis using the IMSS data but the ENOE sample.¹²

1900, 2000, and 2015 Mexican censuses: To calculate general labor market and socioeconomic characteristics for all the Mexican CZs, I use the 1990, 2000, and 2015 population censuses conducted by the Mexican National Institute of Statistics and Geography (*Instituto Nacional de Estadística, Geografía e Informática,* INEGI). I use this information to characterize which CZ observable characteristics are associated with industrial robot import events. Additionally, for the sake of completeness and to facilitate comparisons with existing literature, I compute Bartik-type measures of robot exposure using IFR data.

Mexican economic censuses 2003–2018: I use the Mexican economic census of all firms in the country to compute measures of firm productivity and capital investment within each CZ and sector. The publicly available data are at the municipality–6-digit industry level and can be aggregated to CZ level. The most important limitation of these data is that the frequency is not yearly but every five years.

¹¹Wages in the IMMS dataset are reported as daily taxable income (*salario base de cotización*), which may include various forms of compensation, including paid vacation and end-of-year bonuses. Nonetheless, they may exclude other benefits or compensation not subject to labor income taxation. Notably, wages are both bottom- and top-coded, with approximately 1.3% and 1.7% of observations falling into these categories, respectively. For further details, refer to Puggioni et al. (2022). To ensure consistency, I convert these daily wages to monthly wages by multiplying them by 365 days and dividing them by 12 months.

¹²The ENOE results could be noisier because the survey and administrative data, despite being highly correlated, differ in focus: the administrative data center on firm locations, while the survey data cover worker addresses. Despite my using representative localities, discrepancies may persist because of potential worker location differences (Puggioni et al., 2022, Gutierrez et al., 2023).

3 Institutional Background

3.1 Mexican labor markets and the manufacturing sector

Mexico provides a compelling case for studying the impact of automation on labor markets, particularly within its important manufacturing sector, a key sector for economic development and structural change. This sector is a vital component of Mexico's economy, contributing approximately 17.4% to its GDP and accounting for 30.8% of its total exports and 25.6% of total employment in 2019. Mexico manufactures many products within this sector, including automobiles, aerospace components, electronics, medical devices, textiles, apparel, footwear, and furniture.

The strong integration of Mexico into global value chains, especially with the United States, helps us understand the automation phenomenon. Firms engaged in global value chains face intense competition, particularly from China. They are also typically part of intricate production networks in which precision and reliability, which robots improve, are highly valuable. This explains why industrializing economies, including Mexico, are adopting automation, even with a labor force with relatively lower wages (Artuc et al., 2022). The competitive dynamics in Mexico's industrial landscape can imply that these firms face a higher demand elasticity in the product market, a factor affecting the size of the productivity effect of automation.

I focus the analysis on the period from 2005 to 2020, during which an increasing number of firms started importing and adopting industrial robots between 2014 and 2017 (see Figure 1). The rise of automation coincides with some national economic trends, like decreases in poverty and inequality and increases in income, GDP and foreign direct investment. The unemployment rate in Mexico is low (3% of individuals in the labor force) and remained relatively constant during my period of interest. These patterns already suggest that automation does not have a large negative effect, at least at the aggregate level. But they also suggest caution since automation coincides with potentially positive economic shocks (as noted by Brambilla et al. (2023) for Latin America), which is what my empirical strategy will address.

The sector in which automation has occurred at the largest rate is the manufacturing sector. The most potentially affected workers in this sector are not low–wage workers since 69% of these workers earn 2 to 5 times the minimum wage, and there are fewer low-wage workers than in other sectors (Figure A2, Panel A). The most common occupation in this sector (and thus most likely affected) is industrial operators (65.6%, 69% if formal), who are highly susceptible to automation and are predominantly middle-wage earners.

At the same time, professionals and managers typically earn high wages (see Panel B). Most workers in this industry are young (68% aged 20-44). Note that these groups differ from those most affected in industrialized countries, which are relatively low–wage earners in those countries. In my analysis, I will investigate the effects of automation for each of these more and less affected groups.¹³

3.2 Industrial automation in Mexico

Mexico has experienced a rapid increase in industrial robot adoption in the last decade. Figure 1 illustrates this trend, showing that the stock of operational robots surged at an average annual rate of 40% since 2011, reaching over 45,000 units by 2020. Correspondingly, the value of robot imports has surged since 2006. Simultaneously, the number of municipalities embracing industrial robot imports has risen steeply, starting at 68 in 2006 and reaching 108 by 2018.¹⁴ A driving force behind this rapid automation is the automotive sector and other manufacturing industries, including computer, plastics, basic metals, pharmaceutical, and equipment manufacturing.

Figure A4 shows that, although Mexico's automation growth surpasses that of many nations, its level remains a fraction of that in the US or EU countries. This observation underscores Mexico's early stage of automation, wherein the incremental productivity gains from robot adoption might be comparatively larger. Other Latin American countries, such as Brazil, exhibit substantially lower automation levels that have scarcely changed since 2004.

The contribution of Mexican industry to global value chains increased substantially beginning in the mid to late 2000s, partially explained by the good performance of the automotive sector (Chiquiar, 2019; Iacovone, Muñoz Moreno, Olaberria, & De La Paz Pereira López, 2022). Sectors tightly integrated into global value chains prioritize quality due to the high costs of errors, driving automation adoption even in countries with lower labor costs (Artuc et al., 2022; Robles & Foladori, 2019; Rodrik, 2021). Sectors that adopted robots may be on different trajectories as a result. To address this, I include industry-year fixed effects, therefore only comparing firms within sectors that adopt and do not adopt robots in a given year.

¹³Panel C in Figure A2 indicates that individuals aged 25 to 45 primarily fall into the medium-wage group, with older workers enjoying higher salaries, especially in formal manufacturing. Additional details are available in Table B1, which profiles the workers in each wage category, highlighting the prevalence of industrial operators aged 20–44 among medium-wage earners in manufacturing.

¹⁴This growth is mirrored by the expansion in the number of firms importing robots, which surged from 304 firms in 2006 to 649 in 2018.

Then, why do some firms adopt robots within a specific industry while others do not? Aside from firm-specific shocks, most of this variation comes from supply-side shocks in the robot market. This fact could arise from differences in events at ports or shipping facilities (like accidents, strikes, or changes in regulations), exogenous supplier disruptions (like shortages of electrical and electronics components), or the history of trade ties with countries that supply robots. For example, If a new robot is developed in Japan (or if the price of Japanese robots falls), firms in CZ that have a previous trade connection with Japan are likely to benefit first. These benefits could stem from existing distribution networks, better after-sales service, preferential access, better information, etc.¹⁵ Unfortunately, I do not observe supply chains at the firm level –in fact, my data is at the commuting zone level. So, I cannot exploit this variation. Instead, I will employ several empirical strategies to match CZ as closely as possible and assume that the remaining variation is exogenous from the point of the firm and essentially as good as random, arguing that CZ that are on similar trends have had the same demand shocks but different supply side shocks.

3.2.1 Which CZs import robots?

Previous literature studying automation at the firm level finds that importer firms are positively selected vis-à-vis nonimporters (Acemoglu et al., 2022; Bessen et al., 2020; Koch et al., 2021). Similarly, CZs that receive robot imports differ from those that do not. Table 1 shows that CZs receiving robots tend to be larger in size (population and working-age individuals) and have a higher number of formal workers who earn higher average wages. They also tend to have a larger share of their population living in urban areas and working in manufacturing. Figure A1 shows an increasing trends in formal employment and wages based on the IMSS data, which is common to every wage group and more pronounced in CZs receiving robot imports. This is a characteristic that is correlated with economic and manufacturing growth, and motivates the sample of CZs used in the empirical analysis.

Panel A of Figure 3 presents the estimated coefficients from a linear probability model where the dependent variable is a binary indicator equal to one if a CZ is importing robots and where I simultaneously include many different pre-2006 CZ characteristics. It shows many statistically significant differences between CZs receiving and not receiving robots.

¹⁵The importance of networks and granularity for trade has been documented by recent studies (Chaney, 2014; Gaubert & Itskhoki, 2021).

Various factors positively affect the predicted probability of importation: the workingage population, the share of manufacturing employment (level in 2000 and change in the 1990s), the average formal wage in 2005, and exposure to EU robots in the 1990s. Given these differences between importers and nonimporters, using an empirical strategy that uses all nonimporters as a comparison group would subject my estimates to significant biases: it would be difficult to argue that these very different CZs are on similar trends despite the considerable differences in their economic development.

3.2.2 What can explain the timing of robot imports?

An alternative approach to studying automation's effects involves keeping the CZs that receive at least one industrial robot importation during the analysis period and exploiting the variation in the import timing. Panel B of Figure 3 presents the estimated coefficients from an ordinary least square (OLS) regression where the dependent variable is the first year in which I observe a CZ receiving a robot importation. In this case, no variable except urban population share has a coefficient statistically different from zero. Despite the potential influence of unobservable factors on the timing of the first import, I find no significant and systematic relationship between import timing and relevant observable variables. Additionally, I estimate a linear probability model using the panel of CZs and sectors used in the primary analysis to investigate whether formal employment and wages affect the predicted probability of starting to import robots. Table B3 shows that, after including panel unit and year fixed effects, there is no association between the labor market variables and the event's timing. Given this result, I focus the empirical strategy on comparing CZs that receive robot imports with the ones that have yet to start importing.

4 Empirical Strategy: Event Studies

Identifying the causal effect of robot adoption on jobs and wages presents critical challenges. Given that importing and adopting robots is a decision made by the firms themselves, there is an endogeneity puzzle that necessitates a careful analysis.

On the one hand, imagine a situation where the labor supply diminishes for specific tasks, such as manual and routine ones. For instance, if a company that manufactures electronic gadgets faces a shortage of skilled manual workers who solder electronic components onto circuit boards, attach wiring, and perform quality checks, they may adopt

industrial robots to handle these tasks previously performed by humans. In this scenario, it may appear that robots lead to job reductions, creating a negative bias.

On the other hand, firms could experience positive shocks in product demand. Consequently, they might adopt more robots while increasing the size of their workforce. Similarly, positive financial developments within firms might increase investments in robots and job creation. In such cases, we might erroneously deduce that robots positively affect employment when, in fact, they do not.

The approach that I employ to address this challenge involves leveraging geographical and temporal variations in robot importation while controlling for diverse unobservable factors and shocks at the CZ, 4-digit sector, and year levels. To enhance robustness, I use multiple comparison groups and various methodologies capable of addressing different potential sources of endogeneity.

The main set of strategies that I use is timing-based event study methods. I define an event as the first year in which I observe a CZ receiving industrial robot imports. Then, I can group the CZs into different cohorts by the year of this event. The strategy is to compare the labor market outcomes of CZs receiving industrial robot imports before and after the first importation, using as a comparison group the CZs that have not yet received robot imports but eventually do during the period of analysis. Following Miller (2023), I define the data structure as *timing-based* because I use only ever-treated units and the event dates vary. In this case, the underlying thought experiment is that the event timing is as good as random, implying the traditional assumptions of parallel trends and no anticipation. In this case, I rely on the fact that I find no systematic association between the timing of the importation event and economically relevant observable CZ factors.

Table 2 provides descriptive statistics on the distribution of yearly imports by CZ cohort.¹⁶ There are 104 CZs that receive robots during 2006–2022. Most were already importing robots in 2006, so I exclude these from the sample. I also exclude from the treated sample the observations corresponding to periods after 2019, which are likely affected by the COVID-19 pandemic. After these sample restrictions, I end up with 71 CZs grouped into fourteen cohorts that, on average, imported four times during the analysis period.¹⁷

Two-way fixed effects model: Event-study specification. I first estimate a traditional

¹⁶See Figure A5 in the appendix, which shows a histogram of the events. In addition, Figure A6 shows the raw means of the number of formal workers relative to the number one year before the event for different cohorts and CZs.

¹⁷Note that this does not mean that on average they received four imports (i.e., a firm imported four times). It means that I observe that a CZ received robot imports in four years, regardless of whether there were many or there was just one importation within a year.

two-way fixed effect (TWFE) model. I compare the labor market outcomes of CZs receiving robot imports before and after the event, using the CZs that receive imports earlier or later as the control group. The unit of analysis is a 4-digit sector within a CZ. Therefore, the estimated model is:

$$Y_{smt} = \sum_{\tau=-10, \tau \neq -1}^{10} \beta_{\tau} d_{mt}^{\tau} + \gamma_{ms} + \delta_{str(m)} + \varepsilon_{mst}, \qquad (1)$$

where Y_{mt} is the labor market outcome of industry *s* in CZ *m* in year *t*, d_{mt}^{τ} is dummies indicating the year since the CZ's first robot importation ($\tau = 0$), γ_{ms} is CZ–industry fixed effects, and $\delta_{str(m)}$ is industry–year–region fixed effects. The fixed effects are essential since they capture part of the unobservable factors causing endogeneity. For example, if possible positive demand shocks that importer firms experience happen at the 4-digit industry level, I net out that factor from the analysis.

The β_{τ} parameters are identified by comparing differences in *Y* before and after event τ for treated and not-yet-treated or already-treated units.¹⁸ The identification assumptions underlying this approach are as follows: (i) that there is no systematic difference in how the outcomes evolve between the two groups, except the effect of the treatment (parallel trends assumption), (ii) that units do not anticipate or react to their future treatment status before the event, and (iii) that the effects are homogeneous across cohort groups. This last assumption is particularly important for settings where the analyzed treatment is staggered over time, as it prevents biases arising from comparing later-treated units with earlier-treated ones. Additionally, I cluster the standard errors at the CZ level, representing the most conservative option and corresponding to the level at which I observe robot imports.

New event-study methods. TWFE models can produce biased estimates when there are heterogeneous effects across cohorts under a staggered treatment timing, which is the setting of this study. Therefore, I implement two alternative methods—those of Callaway and Sant'Anna (2021) (CS) and Sun and Abraham (2021) (SA)—that circumvent the issue of negative weighting and the so-called bad comparisons of cohorts. Both methods consist of choosing comparison groups that are not contaminated by the treatment.

The CS method computes multiple 2-by-2 difference-in-differences estimations, avoid-

¹⁸To ensure identification, at least two parameter restrictions are implemented to prevent perfect multicollinearity (Borusyak, Jaravel, & Spiess, 2021; Miller, 2023; Sun & Abraham, 2021). These restrictions involve omitting the event dummy -1 and including end-cap dummy variables at the extremes, such as $d^{\tau \leq 10}mt$ and $d^{\tau \geq 10}mt$.

ing bad comparisons (i.e., not using early-treated observations as a comparison group), and then it aggregates them over event time. The building block of the estimator is ATT(g,t), the average treatment effect on the treated at time t for units starting treatment at time g (cohort). Let us define $D_t = 1$ if it is treated by time t and $D_t = 0$ otherwise (not yet treated). When I use the units not yet treated by time t as the comparison group, the estimand is as follows:

If
$$t \ge g$$
: $ATT(g,t) = \mathbb{E}[Y_t - Y_{g-1}|G = g] - \mathbb{E}[Y_t - Y_{g-1}|D_t = 0, G \ne g]$
If $t < g$: $ATT(g,t) = \mathbb{E}[Y_t - Y_{t-1}|G = g] - \mathbb{E}[Y_t - Y_{t-1}|D_t = 0, G \ne g]$

For post-treatment periods, the method calculates the difference between the outcomes of treated CZs in cohort g and those of not-yet-treated CZs for the period t years after the event compared to the outcomes in the period immediately before the event (period g-1). This procedure implicitly controls for linear trends and accounts for unit and period fixed effects. For pretreatment periods, the method calculates the difference between the outcomes of treated CZs in cohort g and those of not-yet-treated CZs for the period t years compared to those in the period t-1 years before the event.

Then, I can aggregate the multiple ATT(g, t) across event-time to obtain the average effect for units that have been exposed to treatment for exactly *e* time periods (e = t - g):

$$\theta(e) = \sum_{g}^{T} \omega_e(g, t) ATT(g, g + e)$$

=
$$\sum_{g}^{T} \mathbb{1}\{g + e \le T\} P(G = g | G + e \le T) ATT(g, g + e)$$
 (2)

The main difference between the CS and SA methods lies in the comparison group chosen in each. Both methods allow use of the not-yet-treated cohorts as the comparison group. In this context, Sun and Abraham (2021) use the last-treated cohort (cohorts that started importing robots in 2019 or later), while Callaway and Sant'Anna (2021) use all the not-yet-treated cohorts.

Synthetic DD. Synthetic difference-in-differences (SDD) is a hybrid method that combines elements from both the difference-in-differences (DD) and synthetic control (SC) approaches.¹⁹ In an SDD analysis, we use a panel setup, balanced in calendar time, where

¹⁹The method described here follows the framework proposed by Arkhangelsky, Athey, Hirshberg, Imbens, and Wager (2021). Refer to their article for detailed information on estimation, inference procedures, and formal proofs of the estimator's consistency and asymptotic normality.

specific units receive treatment while others remain untreated. The goal is to assess the impact of the treatment by comparing changes in outcomes before and after treatment for both the treated units and a synthetic control group that has the same pretrends. To create the synthetic control group, the method selects units from the never-treated group (CZs that have never received robot imports during the analysis period) and time periods before the event. It assigns unit-specific weights to these untreated units and time-specific weights to the pretreatment periods. SDD capitalizes on the strengths of both the DD and SC methods. Similar to DD models, SDD allows for pretreatment differences in level outcomes between the treated and control units. Conversely, like the SC method, SDD seeks to create a control unit that mitigates the strict requirement of the parallel trends assumption.

As a result, SDD is advantageous because it can overcome common challenges encountered in implementing standard DD and SC methods. It can estimate causal relationships even when pretrends are not strictly parallel in aggregate data, which would violate a key identifying assumption for DD. Furthermore, it does not necessitate that the treated unit fall within the "convex hull" of control units, a condition imposed by the SC method.

Robustness analysis: Alternative methods. I employ a triple-differences (DDD) approach to complement my previous analysis. This method involves comparing labor market outcomes among industries that are more and less exposed to robot adoption within CZs that either receive or do not receive robot imports before and after the first robot importation event. I define more exposed industries as manufacturing industries observed to have a high rate of robot adoption in the IFR data, and low exposed as the rest of non-manufacturing industries.²⁰ The critical innovation of the DDD approach lies in its introducing a third difference by considering different industry exposure levels. This additional layer of differentiation allows me to effectively control for idiosyncratic labor market shocks that might otherwise confound the analysis. These shocks are a primary concern in conventional DD approaches.

I incorporate CZ–year fixed effects into the model to implement this DDD approach. These fixed effects help account for various shocks affecting a CZ labor market in a given year, such as economic downturns, local policy changes, or large-scale industry expan-

²⁰More exposed industries include Transportation, Electronics, Metal products, Chemicals, Plastics, Rubber Energy and power generation, Food, and Beverages. Less exposed industries include all the non-manufacturing industries, excluding wholesale machinery trade. I exclude from the sample the rest of the manufacturing industries, i.e., Textile, Wood, Furniture, Paper, Printing, and Non-metallic mineral products.

sions. By including these fixed effects, I can isolate the impact of robot adoption within industries while minimizing the influence of external shocks or trends that might affect the outcomes. This triple-difference framework enhances the robustness of my analysis and strengthens the causal inference regarding the effects of robot adoption on labor markets.

Finally, I replicate the Bartik instrument strategy outlined in Acemoglu and Restrepo (2020). This strategy involves using an instrumental variable based on exposure to European robot adoption to instrument for exposure to domestic robot adoption. This approach provides an additional layer of validation for my analysis, as it helps isolate plausible exogenous variation in robot adoption. It assumes that European trends reflect reductions in robot costs and technological advancements and do not directly impact Mexican labor markets. Replicating this strategy enables me to compare my findings with results in the existing literature and underscores the significance of my utilizing new industrial robot data. I can improve the measurement of robot adoption and exploit variation within industries. This is crucial as it can be challenging to separate the effects of robots from industry-level shocks, particularly in this context where the automotive industry accounts for most of the phenomenon.

5 Results

5.1 Formal labor market

I begin by examining the reduced-form effects of starting to import industrial robots on CZs' formal employment and average wages for the set of CZs that eventually receive robot imports. In Figure 4 Panel (a), I visually represent the dynamic effects on the logarithmic count of formal workers, defined as individuals affiliated with IMSS in December of each year. In this figure, each dot corresponds to the estimated $\theta(e)$ event-study coefficients, calculated using equation 2, and is accompanied by a 95% confidence interval. Since outcome variables are measured in logs, the point estimates can be interpreted as semi-elasticities.

Consistent with my failing to find observable predictors of the timing of the event for the subset of not-yet-treated CZs, the pre-event coefficients consistently hover close to zero and remain statistically insignificant at the 5% significance level (the uniform confidence interval covers zero for all pre-event years). The average of the coefficients of the periods between one and five years prior to t = 0 is very close to zero (0.005 log points, SE = 0.004), suggesting that employment trended similarly across all CZs before the first robot importation. This finding is reassuring, as it is consistent with the validity of the parallel trends and no-anticipation assumptions.

The number of formal workers in CZs remains relatively stable during the first four years after the initial robot importation, showing an average positive effect of 1.6%, although this effect is not statistically significant at the 5% level. This result suggests that I can rule out significant short-term negative employment effects exceeding -0.7%. Subsequently, the estimated coefficients are still nonnegative. In fact, I find an increase in formal employment averaging 6.9% for five to eight years after the event, a result that is statistically significant at 1%. However, this effect might be explained by compositional changes across CZ cohorts.

The concern about compositional changes arises due to the varying duration of exposure of different cohorts. For instance, we can determine the instantaneous average effect (t = 0) only for the cohort of CZs that started importing robots in 2019. Conversely, for CZs that began importing in 2017, we can assess the effects at event times t = 0, 1, 2. While both cohorts contribute when I compute the coefficient for t = 0, this is not true for t > 0. This situation can introduce confounding dynamics and selective treatment timing among different cohorts if the impacts of robot importation systematically differ.

To mitigate this concern, I balance the sample by including only cohorts exposed to the treatment for a certain minimum number of years and examine dynamic effects within those years. Table 3 presents the results for different balanced samples, ranging from event windows of one to four years around the importation event. Encouragingly, the short-run effects remain nonnegative, allowing me to rule out negative effects within the range of [.4%–3.2%] for formal employment and [.2%–.8%] for average wages, depending on the length of exposure.

On the other hand, it is more difficult to rule out composition effects as an explanation for the positive long-run effects I document. The estimates rely on fewer and fewer units after 4 years have passed. One could in principle restrict the sample to include only units that are observed over a long period of time to estimate these long-run effects. However, there are only a handful of units that are observed for more than 4 years (for example, there are only 2 with a six-year post-period). Thus, the best estimates for these long-run effects come from the synthetic control approach. These estimates show that the effects are, in fact, the largest and most positive for these early adopters (see Figure A13).

This positive employment impact suggests that some of the productivity effects take time to emerge due to learning. The growth might also reflect that the intensity of robot use is increasing over time (for example, CZs that first adopted robots in 2011 continued to adopt more robots in subsequent years). The estimates so far only consider the impact of the first set of robots that are adopted, but in fact automation continues once the CZ starts importing robots. I investigate the effect of the intensity in the next section, and I find that the effects are more positive when the intensity is larger.

Panel (b) of Figure 4 plots the dynamic effects of robot adoption on the logarithmic average real monthly wage for formal workers using the same empirical approach. The patterns mirror those in Panel (a), but the effects seem less pronounced. The real monthly wage does not exhibit differential pretrends before the importation event. Then, there is no statistically significant wage increase during the first four years after the first robot importation, and there is a positive effect after t = 5 of 3% for all industries, which is statistically significant at 5%. If we restrict the sample to robot adopter industries within manufacturing, the effect is still nonnegative (see Table 4).

Another way of understanding whether this null effect is hiding potential effect heterogeneity is to analyze groups that are more exposed, such as manufacturing sector workers. To explore this hypothesis, Table 4 summarizes the effects for various subsamples. Indeed, when I focus the analysis on the manufacturing sector, I find that the shortterm effect is still small and nonnegative but the long-term effect is larger than in the baseline, although more imprecisely estimated and potentially driven by compositional changes. The same is true for other subsamples such as workers in the manufacturing industries that have a large stock of robots according to the IFR data or the sample restricted to a balanced panel in calendar time. A consistent pattern emerges across all these different subgroups, revealing a nonnegative employment effect associated with robot adoption.²¹

Additionally, I explore whether the coefficients are more prominent for groups for which we would anticipate a significant negative effect based on the previous literature. If other groups had notably larger results, this would suggest the presence of omitted variable bias. Table 5 provides the CS model estimates, segmented by various factors such as age, wage, and firm size. This analysis confirms that all groups have the same nonnegative result in the short run. Notably, the long-term employment effects are concentrated among medium-wage and prime-age workers, both groups where industrial operators are more prevalent (see Figure A7 for the dynamic effects). Moreover, it becomes apparent that the results are most pronounced in larger firms with over 50 employees, where

²¹Note that I also consider the inclusion of never-treated CZs in the comparison group. The posttreatment effects are similar to the ones considering only the not-yet-treated sample. However, it exhibits a slightly but statistically significantly differential pretrend.

automation is more likely to occur. Panel D reveals that the employment and wage effects tend to be relatively larger in smaller CZs, which are typically more responsive to labor market shocks (see Figure A8 for further details).

The main results are consistent with Ing and Zhang's (2022) finding for Indonesia of positive employment effects for the manufacturing sector (0.113, SE = 0.038) and wages (0.068, SE = 0.040). These results are similar to this paper's findings, even though these authors consider a more general definition of automation equipment (not only industrial robots). Calì and Presidente's (2021b) employment effect for Indonesia is of the same sign but is larger in magnitude at 0.307 (SE = 0.104). Rodrigo (2021) also does not find a large statistically significant decline on employment for Brazil, and also finds positive effects on wages and productivity. These results suggest that in industrializing economies the effects are positive unlike those found for the US and Europe.

The estimates in this paper differ from those in Brambilla et al. (2023), which finds a negative effect of robot exposure on the number of formal workers in a panel of states in Brazil, Mexico, and Argentina in the periods 2004–2016. A potential source of this difference is that the study uses a shift-share instrument strategy, leveraging industry composition variation across less granular geographic units (states), and exploiting a different source of variation (only within geographic units).

5.2 Robustness checks

5.2.1 Alternative outcome measures and financial crisis

Another concern is related to the measurement of outcomes, which can have measurement errors that would decrease the precision of the estimates since I take the information from December of each year. To explore how sensitive the main results are to this issue, I calculate the labor market outcomes as a simple yearly average of the last month of each quarter (comprising March, June, October, and December). These measures may be a better indicator of what is happening in each year to the extent that they use more available information. Table B4 repeats the main table but changes how I measure the dependent variable. These results closely mirror those derived from the December data exclusively.

A second concern is related to the fact that the observed effects could be driven by firm responses to the 2009 financial crisis. To rule out this potential explanation, I re-estimate the effects excluding from the sample the cohorts that started importing robots in 2009 and 2010. Table B5 shows that the results are very similar when I exclude these cohorts.

5.2.2 Alternative methods

Figure A10 shows the dynamic coefficients of the three event-study models—TWFE, SA, and CS—for the sample of not-yet-but-eventually-treated CZs. In the case of employment, there are no significantly differential pretrends, and the postevent effect is even more pronounced than that estimated under the CS method. In the short term, the number of formal workers increases by between 3.8% and 4.1%, while in the long term, it rises to 10.7% on average. On the other hand, when I focus on average formal wages, the preevent coefficients for the different approaches also lack statistical significance, and the effects follow the same dynamic pattern as those on employment, with more muted and less precise positive results (I can rule out negative wage effects larger than -0.8%).

While the pre-event coefficients derived from the various approaches lack statistical significance, the possibility of a modest pretrend increase in average wages emerges, particularly when I examine a timeframe extending beyond three years before the event. This possibility merits thorough investigation. To assess the significance of these pretrends, I employ a sensitivity analysis method proposed by Rambachan and Roth (2023). This analysis estimates the post-treatment effect under different assumptions of parallel trend violations, demonstrating the extent to which postevent trend differences would need to deviate from the pretrends to render the estimates negative or statistically insignificant. More precisely, I provide confidence intervals that enable the post-treatment deviation from parallel trends to reach a magnitude of up to M times the maximum pretreatment deviation, with M being a variable that can take different values. Figure A11 illustrates the results of this analysis, suggesting that if we assume that the maximum size of the pretrend persists post-treatment, the short-run and long-run estimates for employment and wages remain positive but lack statistical significance (the estimated coefficient for the employment effect in period 6 remains statistically significant when I assume that the pretrend persists but at half its size). The crucial point is that a substantial deviation of the parallel trend assumption would be necessary to yield negative employment and wage estimates.

Another way of addressing the potential violation of parallel pretrends is by controlling for CZ–time-specific idiosyncratic shocks using a DDD model. To do so, I compare the labor market outcomes of more and less exposed industries within CZs before and after the import event between treated and never-treated CZs. I define more exposed industries as those in the manufacturing sector that I observed in the IFR data adopting robots. Less exposed industries are the other, nonmanufacturing industries.²² I further

²²I also exclude from the less exposed industries those in educational service since it is listed as adopting

restrict the sample to CZs with a population of fewer than 500 thousand to improve comparability. Figure 5 shows that, once I control for possible deviations in the employment and wage pretrends using the less exposed group, I still find nonnegative effects on formal employment and wages of similar magnitudes to those in the previous analysis.²³

SDD is an alternative method that directly addresses pretrend differences by construction. When I use this method, I find no statistically significant negative employment or wage effect for any industries or manufacturing sectors, as seen in Figure 6. Most of the employment and wage point estimates are positive but generally have larger confidence intervals than those for the results under the previous methods. For instance, the simple average of the employment effects across CZs is 0.07 (SD=0.30), while the average impact for wages is 0.01 (SD=0.07). Figure A13 shows the difference-in-difference graphs for a specific CZ, where there is a statistically significant positive effect in the number of workers and average wage.

To conclude the analysis, I present the results from the two-stage least squares (2SLS) method with the Bartik instrument of EU automation, using as a measure of domestic robot adoption the change in robot exposure between 2011 and 2015. Figure A14 plots the estimated coefficients and compares them with the results from Acemoglu and Restrepo (2020). Table 8 presents the 2SLS results for multiple outcomes.²⁴ The findings do not show any clear proof of negative effects on jobs or wages, although the standard errors are larger and not the same as those corresponding to the results under previous DD based methods. The reason might be that the Bartik method estimates the effects for a different group of units (those affected by the instrument) and relies on the assumption that if an industry adopts robots, any CZ with firms in that industry adopts robots, even if this is not the case.

Table B7 summarizes the results of all the different methods showing the average estimated coefficients during the pre-event period (averaged across four periods) and the postevent phase, further differentiating the short-run (within four years of the event) from the long-run effects (five to eight years after the event). The main finding across specifications is that there are no substantial negative employment effects of automation. After the first robot importation, there are positive but relatively small effects on the number of formal workers in the short run and larger effects in the long run. This might be consistent with the fact that integrating robots and changes in the job and task landscape

robots in the IFR data but is outside manufacturing. I further exclude the industry of wholesale trade of machinery, which likely includes robot integrators.

²³Figure A12 shows that this conclusion holds if I use 3-digit industry codes.

²⁴Table B6 presents similar results for the change between 2010 and 2015. This version implies imputing robot adoption for the year 2010, given that the IFR data for Mexico are available for years since 2011.

necessitates an adjustment period. Using event studies based on traditional TWFE, either using double or triple differences, I can rule out negative effects larger than -0.4% to -0.8%. Overall, it is encouraging that these alternative approaches support the baseline results.

5.2.3 Other robot adoption measures

Concerns may arise with respect to the relevance of defining the event of interest as a CZ's first year of industrial robot importation. It is essential to consider whether different effects emerge when alternative measures that account for import intensity are considered. To address this, I develop two alternative measures for analysis.

The first one considers the value of the robot imports and defines the event as the year in which the cumulative value of a CZ's robot imports as a share of its total 2005 manufacturing sector payroll surpasses a given threshold. To obtain this measure, I first need to impute the import value for CZs with missing data because, for confidentiality reasons, this information is observable only when more than three firms in the CZ import robots in a given period. Therefore, I impute the import values using the mean of each year-state-country supplier cell. Panel (a) of Figure A9 in the appendix shows that the imputed measure has a mean and variance similar to those of the nonimputed measure. Panel (b) compares the distribution of CZs across the time passed since the event as originally defined and the event as defined by means of the cumulative import values with a 50% threshold. Under the new definition, there are fewer CZs treated for fewer periods. Figure A15 shows the CS estimates for events based on the new definition under increasing thresholds. All the estimated coefficients for formal employment and wages are positive, regardless of the threshold. In fact, the average of the long-term coefficients is statistically significant at 1% when I use thresholds larger than 40% (i.e., the cumulative value of robot imports is larger than 40% of the total manufacturing sector wage bill). Table 6 summarizes these results.

A second possibility is to use the cumulative number of yearly imports at the CZ level. Since the Callaway and Sant'Anna (2021) method does not allow me to consider the treatment intensity, I implement this measure using the TWFE model, interacting the event-time dummies with the cumulative number of yearly imports. Table 6 shows that the employment and wage effects are similar to the baseline estimates described in the previous section.

5.3 Other margins of adjustment

To evaluate whether firms and workers use alternative margins of adjustment, I conduct an analysis of its effects on labor informality, unemployment, and labor force participation using ENOE data. Table 7 presents the results of our primary specification, using formal and informal employment measures obtained from ENOE as dependent variables. These results are compared with point estimates derived from the IMSS sample, restricting the sample to CZs for which the ENOE data are representative. It is worth noting that ENOE is representative for only a limited number of major cities. Consequently, the estimates regarding the number of formal workers, when compared to those calculated from the IMSS data, exhibit differences in precision, with the latter being less statistically robust but qualitatively consistent (showing a positive short-term effect and a more pronounced long-term effect).

Notably, our findings reveal a short-term decrease in the number of workers in the informal sector, followed by a long-term increase, potentially attributable to general-equilibrium effects. A similar pattern emerges in the case of unemployment and labor force participation. Intriguingly, there is a concurrent rise in the urban population, suggesting that there might be rural-to-urban migration.

5.3.1 Ruling out potential explanations: Plant openings

In dissecting the underlying factors behind the observed positive employment effects beyond the anticipated gains in productivity from robot adoption, I delve into the prospect that firms, buoyed by robust demand shocks, opt for expansion and open new facility establishments, potentially internationally. This sequence could subsequently trigger robot incorporation and a surge in the workforce.

To assess this, I analyze the sensitivity of the estimated robot–employment relationship to my factoring in a measure of plant openings. Leveraging the 2018 economic census, I utilize the variable tracking the commencement year of operations. This enables the calculation of establishment openings for surviving firms. I focus this analysis on specific industries and firm sizes, particularly on industries substantially exposed to automation and larger firms.

Importantly, these segments are more likely to have persisted until 2018, rendering the measure more reliable. The core point estimates remain largely unaltered upon my including this measure into the CS model (see Figure A16). Furthermore, I extend this exploration to directly evaluate the impact of robots on plant openings, finding no dis-

cernible association (see Figure 7).

In essence, this comprehensive exploration effectively narrows down the plausible drivers of potential positive employment effects, bolstering the conjecture that heightened productivity triggered by robot adoption was a significant catalyst. This fact lends further weight to the argument that the positive employment outcomes observed are rooted primarily in productivity enhancements brought about by adopting robots.

6 Mechanisms and Alternative Explanations

To analyze the effects of industrial robot adoption on productivity, output, the labor share, and capital investment, I use the Mexican economic censuses (2003, 2008, 2013, 2018). These censuses are taken every five years, which limits my ability to explore the dynamic effects of robot adoption. For this analysis, I estimate the following TWFE model:

$$Y_{smt} = \beta D_{mt}^{post} + \gamma_{sm} + \delta_{st} + \epsilon_{smt},\tag{3}$$

where Y_{smt} represents the aggregated outcomes of firms at the six-digit sector s in CZ m in census year t and D_{mt}^{post} is an indicator variable that takes the value of one if CZ m starts or has already started to import robots in census year t. This model allows me to control for idiosyncratic shocks at a much more granular (six-digit) sector level than in the analysis with the IMSS data, where I can control for shocks only at the four-digit sector level.²⁵ γ_{sm} is CZ × six-digit sector fixed effects, which capture non-time-varying factors, such as product or labor market conditions. δ_{st} is six-digit sector × census year fixed effects, which capture the common trends across sectors over time. For example, some sectors may have regional policies or may experience demand shocks due to global competition or trade liberalization that affect all firms in the sector over time. By including these fixed effects, I can control for important confounding factors that may affect the relationship between robot adoption and the aggregated firm outcomes.

Table 9 presents the results and is divided into two panels: Panel (a) shows the results for all industries, and Panel (b) shows the results for a subset of manufacturing sectors that are more likely to be affected by automation (metal, machinery and equipment, plas-

²⁵One example of a six-digit sector within the automotive manufacturing sector is 336111, automobile manufacturing. This sector comprises establishments primarily engaged in one or more of the following: (1) manufacturing complete automobiles (i.e., body and chassis or unibody); (2) producing automobile chassis only; (3) manufacturing automobile bodies only; or (4) manufacturing and/or assembling automobile sub-assemblies. This sector is part of 3361, motor vehicle manufacturing, which is part of 336, transportation equipment manufacturing.

tic and rubber, power generation, and transportation).²⁶ The dependent variables are the number of firms, labor share, capital per worker, ln gross value added (GVA), GVA per worker, input cost, total payroll, total fixed assets, number of workers, number of workers in production and sales, number of workers in administration, hours worked, hours worked in production and sales, and hours worked in administration.

For all industries, industrial robot adoption has a positive and significant effect on GVA and the input cost. This suggests that robot adoption may increase the productivity of firms in Mexico. However, robot adoption has no significant effect on the labor share, capital per worker, payroll, fixed assets, or employment. This suggests that robot adoption does not affect the distribution of income between labor and capital, or firm size. However, this could be result of the lack of power given the level of aggregation and the availability of observations for only few periods. The sign of the labor share is negative, and the magnitude is -1.1%, consistent with results derived from task-based frameworks for considering automation effects (Acemoglu & Restrepo, 2020).

For the subset of manufacturing sectors that are more likely to be affected by automation, industrial robot adoption has a positive and significant effect on GVA per worker, input cost, payroll, fixed assets, and capital per worker. This suggests that robot adoption increases the productivity, efficiency, wages, and investment of firms in these sectors. However, robot adoption has no significant effect on the labor share (although the coefficient is negative and larger in magnitude than that for all industries). This suggests that robot adoption does not affect the distribution of income between labor and capital.

I find that robot adoption has a nonnegative effect not only on employment but also on hours worked. Across all industries, the effect is generally small and lacks statistical significance. However, when I narrow my focus to the manufacturing sector, the effects become more substantial and statistically significant within the balanced sample: employment shows an increase of 18.1%, while hours worked increase by 19.4%. These effects can be attributed to increased production and sales workers rather than administrative workers.²⁷.

For the other manufacturing sectors, the effects are either not significant or much smaller than the effects found for the manufacturing sectors that are more likely to be

²⁶Tables B8 and B9 present the same results for the balanced panel in calendar time. The findings are similar to those from the unbalanced panel.

²⁷The differences in the effects on production and administrative workers have been investigated by Dixon et al. (2019). They used firm-level data from Canada and found that adopting robots significantly reduces the number of managers and administrative staff, and increases productivity and wages. According to their hypothesis, robots enable companies to enhance their product and service quality, which, in turn, requires more flexible and adaptable management practices.

affected by automation. However, Panel (a) of Table 10 presents some exceptions, showing evidence for sectors such the wood, paper and chemicals industries of an increase in capital per worker (3.2%) and a reduction in employment of production and administrative workers. This finding could suggest a certain degree of reallocation of workers across sectors, depending on how much they automate and how large the productivity channel is. I find no effects for manufacturing industries with very low levels of robot adoption, such as the textile sector (see Panel (b)).

For the nonmanufacturing sectors (Panel (c) in Table 10), industrial robot adoption has a positive and significant effect on GVA, input cost, number of workers, and hours worked. However, these effects are much smaller than those for the manufacturing sectors more likely to be affected by automation. These outcomes may indicate the presence of broader, general-equilibrium effects stemming from robot adoption. Such effects can include shifts in demand or supply dynamics across various sectors, including construction, wholesale and retail trade, and manufacturing, ultimately highlighting the interconnected nature of these industries.

These findings align with the results in the literature on industrial robot adoption at the firm level, highlighting its positive impact on productivity and employment, as evidenced by significant increases in GVA per worker and number of workers (for example, Acemoglu et al. (2023) for the Netherlands, Acemoglu et al. (2022) for the US, Acemoglu et al. (2021) for France, Koch et al. (2021) for Spain, Dixon et al. (2019) for Canada, and Humlum (2020) for Denmark). These aggregate estimates might hide effects within industry since robot adopters may expand at the expense of nonadopter competitors (Acemoglu et al., 2023). However, I still find zero or positive effects on aggregate employment.

7 Discussion

7.1 Theoretical framework

To understand how industrial robot adoption may impact labor markets differently in developing countries, I summarize the key findings of a simple, competitive labor markets model in the style of Acemoglu and Restrepo (2018) where automation allows robots to substitute for workers in some tasks. Acemoglu and Restrepo (2018) develop a task-based model in which robots and human labor compete to complete various tasks. In this model, the effects of robots on wages and employment are ambiguous.

Consider a perfectly competitive economy with two sectors j: (i) manufacturing (j =

1) and (ii) the rest of the sectors (j = 2), potentially including the informal sector in the Mexican context. Production at j = 2 uses only workers, such that one unit of labor corresponds to one unit of output.

The representative firm in sector 1 produces a final good Q by combining a continuum unit measure of tasks, which are completed by robots and human workers. I assume that the markets for both inputs are perfectly competitive. While workers have identical productivity in all tasks, robots are as productive as workers on tasks in the range [0, I]but have zero productivity at tasks (I, 1]. Robots can be purchased in an international competitive market at constant rental rate r. I am primarily interested in the equilibrium effects of I, that is, increased automation that allows robots to perform tasks that previously required labor.

Good 2 can be produced internally, as noted above, or imported at constant price p_2 . I set this good as the numeraire in the economy: $p_2 \equiv 1$.

The manufacturing good is exported, facing isoelastic international demand:

$$Q(P_1) = \kappa P_1^{-\sigma}$$

where the constant κ is exogenous.

Turning to individual preferences, I assume that all individuals *i* consume only good 2, and the amount they consume is equal to the wage they receive, as the price is normalized to one. Their indirect utility is a function of their choice of employment sector (j = 1 or 2) and the corresponding wage, and is given by:

$$U_{ij} = \beta ln(w_j) + \epsilon_{ij},$$

where ϵ_{ij} is i.i.d. extreme value type 1 errors, indicating that workers may have a preference for one sector over the other. I denote the equilibrium wage in each sector by w_j and the cost of robots by r. Wages can be different across sectors because workers have idiosyncratic preferences for where they work. Given the assumption that production of good 2 requires one unit of labor, and that the market for good 2 is perfect competitive, it follows that $w_2 = 1$.

I assume that the cost of robots is lower than the cost of labor $r < w_1$, such that all tasks in [0, I] are performed by robots while tasks in (I, 1] are completed by human workers, where *I* represents the extent of automation. Therefore, the cost per unit is:

$$Ir + (1 - I)w_1 = P_1.$$

We obtain the labor demand by plugging the expression for P_1 into the aggregate demand for good 1:

$$Q(P_1) = \kappa [Ir + (1 - I)w_1]^{-\sigma}$$
$$LD_1 = (1 - I)\kappa [Ir + (1 - I)w_1]^{-\sigma}$$

Each worker with wage w_j and extreme value type 1 heterogeneity ϵ_{ij} evaluates the utility for each option and chooses the highest utility. Consequently, the aggregate labor supply of the manufacturing sector is as follows:

$$LS_1(w_1) = \frac{w_1^{\beta}}{w_1^{\beta} + w_2^{\beta}} = \frac{w_1^{\beta}}{w_1^{\beta} + 1} = \frac{1}{1 + w_1^{-\beta}},$$

where β corresponds to the elasticity of supply.

Equilibrium in the labor market is given by $LS_1(w_1^*) = LD_1(w_1^*)$, such that:

$$\frac{1}{1+w_1^{-\beta}} = (1-I)\kappa[Ir + (1-I)w_1]^{-\sigma}.$$

In equilibrium, the elasticity of labor supply β affects the slope of the labor supply curve, affecting the size of the employment and wage effect but not its direction. When β is relatively low, i.e., inelastic, the effects of shifts in labor demand on employment are attenuated and wages exacerbated.

I am more interested in the direction of the effect governed by the parameter σ of labor demand, representing the price elasticity of aggregate demand for manufacturing products. Suppose that σ is relatively high. In this case, the labor demand curve shifts to the right, and there is a positive effect on employment and wages. Figure 8 graphically shows how the equilibrium depends on the parameters in a standard supply and demand model.

I next analyze how labor demand changes when the extent of automation marginally increases.

Writing the labor demand function in logs:

$$ln(LD_1(w_1)) = ln(1-I) + ln(\kappa) - \sigma ln[Ir + (1-I)w_1]$$
(4)

Differentiating 4 with respect to *I* leads to the following expression;

$$\frac{\partial LD_1(w_1)}{\partial I} = -\frac{1}{1-I} - \sigma \frac{(r-w_1)}{Ir + (1-I)w_1}.$$

Given the assumption $r < w_1$, I can rewrite this as:

$$\frac{\partial LD_1(w_1)}{\partial I} = \underbrace{-\frac{1}{1-I}}_{\text{Displacement effect}} + \underbrace{\frac{w_1 - r}{Ir + (1-I)w_1}}_{\text{Productivity effect}}.$$

According to this result, extending automation has an ambiguous effect on labor demand. On the one hand, there is a displacement effect, which is always negative, $-\frac{1}{1-I} < 0.^{28}$ On the other hand, there is a productivity or scale effect that is positive under the assumption that $r < w_1$. This latter effect crucially depends on the size of the elasticity of labor demand σ . The larger σ is, the larger the productivity effect, which may more than compensate for the negative displacement effect.

Automation drives the substitution of labor with capital when capital can perform specific tasks more efficiently at the margin. The productivity effect rises because of a reduction in production costs, which translates into a reduction in prices for goods, increasing real income for consumers and consumer demand. In turn, this increases labor demand within nonautomated tasks or sectors yet to undergo automation. This is why, if consumers are more price sensitive, there could be a larger increase in demand and the productivity effect might offset any adverse effect.

7.2 Why might the productivity effect be larger in industrializing than in rich countries?

To understand whether the positive productivity effects of automation are more likely to be larger in industrializing economies such as Mexico, it is helpful to analyze where the productivity gains from robot adoption come from. As explained in Section 7.1, the productivity gains from automation do not result from capital and labor becoming more productive in tasks that they were already performing—they are a consequence of firms' abil-

²⁸In Acemoglu and Restrepo (2020), the formula for the cost savings from using robots in labor market c is in terms of relative prices: $\pi_c = 1 - \frac{\gamma_L}{W_c} \frac{R_c^M}{\gamma_M}$, where γ is factor productivity and W and R^M the prices of labor and capital, respectively. The greater the productivity of capital in automated tasks relative to the rental rate of capital, and the smaller the productivity of labor in these tasks relative to wages, the greater are the productivity gains from automation.

ity to use cheaper capital in functions previously performed by human workers. Therefore, the productivity effect that arises is proportional to the cost savings.

However, the productivity effect also depends on the size of the elasticity of labor demand σ . If the elasticity is large enough, with all else held constant, the productivity effects could offset or even more than compensate for the displacement effect and increase labor demand. Are there reasons why the elasticity of demand for goods in Mexico be different from that in the US?

Automation in Mexico happens mostly in specific manufacturing industries that are generally concentrated on labor-intensive tasks (assembly) and producing tradable goods for export. Firms in these sectors often face foreign competition, leading to price reductions through automation, creating a competitive advantage and boosting output.²⁹ On the other hand, US firms might face low competition from abroad since they could be more interested in satisfying domestic demand and consumers may prefer US goods. Consumers in the US also have higher earnings, so they might be less price-sensitive than Mexican consumers.

Alternative explanations are related to different type of industrial robots being adopted depending on the automation stage. Acemoglu and Restrepo (2020) maintain that the predominant negative displacement effect in the US aligns with the prevalence of low-productivity automation technology (referred to as *so-so automation*), which is sufficiently productive that firms adopt it in the presence of factors such as tax code distortions but not inherently more cost-effective than the processes that it supplants.³⁰ Consistent with this idea, Graetz and Michaels (2018) and Calì and Presidente (2021b) suggest that the positive productivity effect could be larger in developing countries because of diminishing marginal gains from automation. This implies that, as robot density increases, the corresponding productivity gains become increasingly smaller. Furthermore, negative displacement effects might be smaller in Mexico than in the US, given that the labor displaced and early-stage automation might have higher productivity.

²⁹For example, Harasztosi and Lindner (2019) find that the tradable sector and exporting firms have larger responses to minimum wage increases because they are more exposed to foreign competitors and have a higher price elasticity of demand.

³⁰In the context of so-so automation, the technology's productivity only marginally surpasses that of labor $(\frac{\gamma_L}{W_c} \approx \frac{R_c^M}{\gamma_M})$. Nonetheless, in Mexico, the likelihood of $\frac{\gamma_M}{R_c^M} \gg \frac{\gamma_L}{W_c}$ is heightened, particularly considering that so-so automation tends to emerge at later industrialization stages.

8 Conclusion

This study sheds light on the impact of industrial robot adoption on labor markets in developing countries, with Mexico serving as a compelling case study. The findings offer evidence to support an optimistic narrative: the rapid adoption of robots in Mexico over the last decade has not yielded substantial negative effects on formal employment or average real formal wages. Instead, it sets a favorable transitional phase, with far-reaching implications. While the immediate impact may appear modest, there is evidence of a potential positive effect in the long run, underlining the gradual labor market adjustments linked to automation.

These findings suggest possible policy considerations for industrializing countries such as Mexico. Policymakers can explore the potential of promoting automation to increase productivity and growth, especially in the tradable manufacturing sector. However, the equitable distribution of these benefits is a crucial aspect that needs attention. Facilitating technology transfer from automating to non-automating sectors through partnerships and collaborations could be a viable strategy. Policymakers can adopt a forwardthinking approach, anticipating a future where automation plays a significant role in the economy. This approach could involve planning for investments in education and training for jobs in automated industries, developing infrastructure to support automation, and formulating regulations to guide the ethical use of these technologies. Adopting such a long-term perspective could ensure a smooth and beneficial transition to a more automated economy.

In conclusion, this study contributes to the emerging body of evidence on automation's impact on industrializing economies and suggests promising avenues for future research. Further investigations using firm- and establishment-level data can provide a deeper understanding of the mechanisms and dynamics underlying the observed effects. Exploring worker-level effects will enable a more nuanced assessment of reallocation effects and of which workers might be beneficiaries and which might be adversely affected, offering valuable insights for public policy interventions. Comparative studies across regions and countries hold the potential to reveal the nuanced impact of automation across different contexts, facilitating more targeted and effective policy responses. As we navigate the transformative era of automation, continued research in these directions will be essential to inform evidence-based policymaking and ensure inclusive and sustainable economic development.

References

- Acemoglu, D., Anderson, G., Beede, D., Buffington, C., Childress, E., Dinlersoz, E., ...
 Zolas, N. J. (2022). Automation and the Workforce: A Firm-Level View from the 2019 Annual Business Survey. *SSRN Electronic Journal*. doi: 10.2139/ssrn.4282509
- Acemoglu, D., Koster, H., & Ozgen, C. (2023). Robots and Workers: Evidence from the Netherlands. *SSRN Electronic Journal*. doi: 10.2139/ssrn.4387930
- Acemoglu, D., Lelarge, C., & Restrepo, P. (2021). Competing with Robots: Firm-Level Evidence from France. *SSRN Electronic Journal*. doi: 10.2139/ssrn.3535328
- Acemoglu, D., & Restrepo, P. (2018). The race between man and machine: Implications of technology for growth, factor shares, and employment. *American Economic Review*, 108(6), 1488–1542. doi: 10.1257/aer.20160696
- Acemoglu, D., & Restrepo, P. (2019). Automation and new tasks: How technology displaces and reinstates labor. *Journal of Economic Perspectives*, 33(2), 3–30. doi: 10.1257/jep.33.2.3
- Acemoglu, D., & Restrepo, P. (2020). Robots and jobs: Evidence from us labor markets. *Journal of Political Economy*, 128(6), 2188–2244. doi: 10.1086/705716
- Acemoglu, D., & Restrepo, P. (2022a). Tasks, Automation, and the Rise in U.S. Wage Inequality. *Econometrica*, 90(5), 1973–2016. doi: 10.3982/ecta19815
- Acemoglu, D., & Restrepo, P. (2022b). Tasks, Automation, and the Rise in U.S. Wage Inequality. *Econometrica*, 90(5), 1973–2016. doi: 10.3982/ecta19815
- Aghion, P., Antonin, C., Bunel, S., & Jaravel, X. (2022). Modern manufacturing capital, labor demand, and product market dynamics: Evidence from France Working Paper.
- Arkhangelsky, D., Athey, S., Hirshberg, D. A., Imbens, G. W., & Wager, S. (2021). Synthetic Difference-in-Differences. *American Economic Review*, 111(12), 4088–4118. doi: 10 .1257/aer.20190159
- Arntz, M., Gregory, T., & Zierahn, U. (2019). Digitalization and the future of work: Macroeconomic consequences.
- Artuc, E., Bastos, P., Copestake, A., & Rijkers, B. (2022). *Robots and Trade: Implications for Developing Countries* (No. 2018). doi: 10.4324/9781003275534-8
- Artuc, E., Christiaensen, L., & Winkler, H. (2019). Does Automation in Rich Countries Hurt Developing Ones? Evidence from the U.S. and Mexico. Does Automation in Rich Countries Hurt Developing Ones? Evidence from the U.S. and Mexico(February). doi: 10.1596/31425
- Atkin, D. (2016). Endogenous skill acquisition and export manufacturing in Mexico.
American Economic Review, 106(8), 2046–2085. doi: 10.1257/aer.20120901

- Autor, D. H., & Dorn, D. (2013). The growth of low-skill service jobs and the polarization of the US Labor Market. *American Economic Review*, 103(5), 1553–1597. doi: 10.1257/ aer.103.5.1553
- Bessen, J., Goos, M., Salomons, A., & van den Berge, W. (2020). Firm-Level Automation: Evidence from the Netherlands. *AEA Papers and Proceedings*, 110, 389–393. doi: 10.1257/pandp.20201004
- Borusyak, K., Jaravel, X., & Spiess, J. (2021). Revisiting Event Study Designs: Robust and Efficient Estimation. (2014), 1–85. Retrieved from http://arxiv.org/abs/2108.12419
- Brambilla, I., César, A., Falcone, G., & Gasparini, L. (2023). The impact of robots in Latin America: Evidence from local labor markets. *World Development*, 170, 106271. Retrieved from https://doi.org/10.1016/j.worlddev.2023.106271 doi: 10.1016/j.worlddev.2023.106271
- Calì, M., & Presidente, G. (2021a). Automation and manufacturing performance in a developing country. *Oxford Martin School, Oxford University*, 1–58.
- Calì, M., & Presidente, G. (2021b). Robots For Economic Development. *IDEAS Working Paper Series from RePEc*, 1–78.
- Callaway, B., & Sant'Anna, P. H. (2021). Difference-in-Differences with multiple time periods. *Journal of Econometrics*, 225(2), 200–230. Retrieved from https://doi.org/10.1016/j.jeconom.2020.12.001 doi: 10.1016/j.jeconom.2020.12.001
- Chaney, T. (2014). The network structure of international trade. *American Economic Review*, 104(11), 3600–3634. doi: 10.1257/aer.104.11.3600
- Chen, C., & Frey, C. B. (2021). Oxford Martin Working Paper Series on Economic and Technological Change Automation or Globalization? The Impacts of Robots and Chinese Imports on Jobs in the United Kingdom Automation or Globalization? The Impacts of Robots and Chinese Imports on Jobs in the United Kingdom *. Retrieved from https://www.oxfordmartin.ox.ac.uk/future-of-work/
- Chiacchio, F., Petropoulos, G., & Pichler, D. (2018). The impact of industrial robots on EU employment and wages. Retrieved from http://bruegel.org/2017/12/ the-growing-presence-of-robots-in-eu-industries/.
- Chiquiar, D. (2019). Working Papers Global Value Chains in Mexico : A Historical Perspective Global Value Chains in M exico : A Historical Per s p ecti v e *.
- Dauth, W., Findeisen, S., Suedekum, J., & Woessner, N. (2021). The Adjustment of Labor Markets to Robots. *Journal of the European Economic Association*, 19(6), 3104–3153. doi: 10.1093/jeea/jvab012

- Dauth, W., S Findeisen, Suedekum, J., & Woessner, N. (2018). Adjusting to robots: Workerlevel evidence", opportunity and inclusive growth. *Institute Working Paper Federal Reserve Bank of Minneapolis.*, 13.
- Dixon, J., Hong, B., & Wu, L. (2019). The Employment Consequences of Robots: Firm-Level Evidence. *SSRN Electronic Journal*. doi: 10.2139/ssrn.3422581
- Faber, M. (2020). Robots and reshoring: Evidence from Mexican labor markets. *Journal of International Economics*, 127, 103384. Retrieved from https://doi.org/ 10.1016/j.jinteco.2020.103384 doi: 10.1016/j.jinteco.2020.103384
- Frey, C. B., & Osborne, M. A. (2017). The future of employment: How susceptible are jobs to computerisation? *Technological Forecasting and Social Change*, 114, 254–280. Retrieved from http://dx.doi.org/10.1016/j.techfore.2016.08.019 doi: 10.1016/j.techfore.2016.08.019
- Gaubert, C., & Itskhoki, O. (2021). Granular comparative advantage. *Journal of Political Economy*, 129(3), 871–939. doi: 10.1086/712444
- Graetz, G., & Michaels, G. (2018). The Review of Economics and Statistics. *The Review of Economics and Statistics*, 100(5), 753–768. doi: 10.1162/rest
- Humlum, A. (2020). Robot Adoption and Labor Market Dynamics. *Unpublished Manuscript*, 1–78.
- Iacovone, L., Muñoz Moreno, R., Olaberria, E., & De La Paz Pereira López, M. (2022). *Productivity Growth in Mexico*. World Bank Group.
- IFR. (2022). World Robotics Report: "All-Time High" with Half a Million Robots Installed in one Year.
- Ing, L. Y., & Zhang, R. (2022). Automation in Indonesia: Productivity, Quality, and Employment. doi: 10.4324/9781003275534-9
- Koch, M., Manuylov, I., & Smolka, M. (2021). Robots and Firms. *Economic Journal*, 131(638), 2553–2584. doi: 10.1093/ej/ueab009
- Kugler, A. D., Kugler, M., Ripani, L., & Rodrigo, R. (2021). U.S. Robots and Their Impacts in the Tropics: Evidence from Colombian Labor Markets. SSRN Electronic Journal. doi: 10.2139/ssrn.3723271
- Miller, D. L. (2023). An Introductory Guide to Event Study Models. *Journal of Economic Perspectives*, 37(2), 203–230. doi: 10.1257/jep.37.2.203
- Porta, R. L., & Shleifer, A. (2008). The unofficial economy and economic development. *Brookings Papers on Economic Activity*, 2008(2), 275–352. doi: 10.2139/ssrn.1304760
- Porta, R. L., & Shleifer, A. (2014). Informality and Development. *Journal of Economic Perspectives*, 28(3), 109–126.
- Puggioni, D., Calderón, M., Cebreros Zurita, A., Fernández Bujanda, L., Inguanzo

González, J. A., & Jaume, D. (2022). Inequality, income dynamics, and worker transitions: The case of Mexico. *Quantitative Economics*, *13*(4), 1669–1705. doi: 10.3982/qe1863

- Rambachan, A., & Roth, J. (2023). A More Credible Approach to Parallel Trends. *Review* of *Economic Studies*(February), 2555–2591. doi: 10.1093/restud/rdad018
- Robles, R., & Foladori, G. (2019). Una revisión histórica de la automatización de la minería en México. *Revista Problemas del Desarrollo*, 50(197), 157–180.
- Rodrigo, R. (2021). Robot Adoption, Organizational Capital, and the Productivity Paradox. *Working Paper*(November), 1–50.
- Rodrik, D. (2021). New Technologies, Global Value Chains, and the Developing Economies. *SSRN Electronic Journal*. doi: 10.2139/ssrn.3338636
- Stemmler, H. (2020). Automated Deindustrialization: How Global Robotization Affects Emerging Economies - Evidence from Brazil. (382), 58. Retrieved from https:// drive.google.com/file/d/1H2C4kSe1EvA35xJrHX1AJSNwM6vQMJux/ view
- Sun, L., & Abraham, S. (2021). Estimating dynamic treatment effects in event studies with heterogeneous treatment effects. *Journal of Econometrics*, 225(2), 175–199. Retrieved from https://doi.org/10.1016/j.jeconom.2020.09.006 doi: 10.1016/ j.jeconom.2020.09.006
- Timmer, M., de Vries, G., & de Vries, K. (2015). Patterns of structural change in developing countries. In J. Weiss & M. Tribe (Eds.), *Routledge handbook of industry and development* (pp. 65–83). Routledge.
- Ulyssea, G. (2018). Firms, informality, and development: Theory and evidence from Brazil. *American Economic Review*, *108*(8), 2015–2047. doi: 10.1257/aer.20141745
- WCO. (2013). HS Classification Handbook (No. November). Retrieved from http:// harmonizedsystem.wcoomdpublications.org

Figures and tables



FIGURE 1: ROBOT ADOPTION IN MEXICO

(b) Flow of industrial robots from IFR data

Notes: Panel (a) of Figure 1 depicts the flow evolution of the number of Mexican municipalities (similar to US counties) importing robots between 2006 and 2022. Panel (b) shows the evolution of the flow of industrial robots adopted in Mexico according to IFR data. *Data sources*: Mexican customs records from Secretaría de Economía, own calculations based on IFR World Robotics Database.

FIGURE 2: AUTOMATION GEOGRAPHY: ROBOT IMPORTS ACROSS COM-MUTING ZONES



(b) Including large plants in the transportation manufacturing subsector

Notes: Figure 2 depicts the geographic distribution of robot imports across commuting zones. Darker shades indicate commuting zones with any robot imports during 2006–2022. The location of importation refers to the exact municipality where purchasing companies are legally registered, as determined by their fiscal address. This registration location may not necessarily coincide with the physical location of the production plant. Red dots correspond to the location of 991 large plants, defined as establishments with more than 250 employees, in transportation equipment manufacturing (NAICS 336). *Data sources*: Mexican customs records from Secretaría de Economía. National Statistical Directory of Economic Units (DENUE, 2020).

FIGURE 3: PREDICTING WHO IMPORTS ROBOTS AND WHEN









Notes: Panel (a) in Figure 3 displays the estimated OLS coefficients along with the corresponding 95% confidence interval of a linear probability model. In this model, the dependent variable indicates whether the CZ received robot imports between 2006 and 2022. Panel (b) illustrates the estimated OLS coefficients of a model with the year of the first robot importation event as the dependent variable. CZs that began importing robots in 2006 are excluded from the sample. The standard errors are clustered at the CZ level. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data. Mexican 1990 and 2000 censuses.

FIGURE 4: ROBOT ADOPTION AND FORMAL LABOR MARKET





Notes: Figure 4 plots the estimated $\theta(e)$ event-study coefficients using equation 2 (Callaway and Sant'Anna (2021) method). The outcome variable is, in turn, the log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican consumer price index (CPI) (Panel (b)). The event is defined as a first-time industrial robot importation. The vertical lines reflect the 95% pointwise confidence intervals, while the shadowed area reflects the 95% uniform confidence intervals, which are robust to multiple hypothesis testing. Horizontal lines show the simple average estimated coefficient for different periods: pre-event refers to years -5 to -1; postevent short run refers to years 0 to 4; long-run refers to years 5 to 8. The clustering of standard errors is at the CZ level. Sample of not-yet-but-eventually-treated CZs for 2005–2019, including cohorts that started importing robots during 2007–2022. The unit of observation is CZ–4-digit industry (N=100,619; 71 CZs). *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

FIGURE 5: TWFE MODELS: DD & DDD



(a) DD estimates. CZ–4-dig. industry level



Notes: Figure 5 displays the estimated event-study coefficients obtained using equation 2, a TWFE model. In Panel (a), the results of the difference-in-differences specification are presented, showing overlapping estimates from two different subsamples: "high robot adopters", comprising manufacturing industries with higher rates of robot adoption observed in the IFR data, and "low robot adopters", which include the remaining nonmanufacturing industries. Panel (b) shows estimates from the triple difference-in-differences specification, where the third difference is introduced by contrasting high and low robot adopters. The outcome variable is, in turn, the log number of formal workers and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI. The event is defined as a first-time industrial robot importation. The clustering of standard errors is at the CZ level. Sample of CZs with fewer than 500 thousands inhabitants for the period 2005–2019, including cohorts that started importing robots during 2007–2022. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

Figure 6: SDD ESTIMATES FOR EACH CZ FROM COHORTS 2010–2016



(a) Sample: All industries

Notes: Figure 6 exhibits the estimated treatment effects using the synthetic difference-in-difference model as outlined by Arkhangelsky et al. (2021). In this figure, each dot represents the estimated effect for individual CZs belonging to cohorts from 2010 to 2016, and the gray bars indicate the corresponding 95% confidence intervals. The comparison group for these estimates is derived from the pool of all CZs that have never been treated, and it employs optimal weighting of units and pre-event periods in its construction. The dataset comprises a panel of CZs, balanced in calendar time, spanning from 2005 to 2019. Panel (a) presents the results for the aggregate of all industries, while Panel (b) focuses specifically on the results for the sample of aggregate manufacturing industries. The outcome variables are the logarithm of the number of formal workers and the logarithm of the average real monthly wage for formal workers, adjusted to constant 2010 pesos using Mexican CPI. The event is defined as a first-time industrial robot importation. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.





A. Number of large plant openings conditional on the plants' surviving until 2018



B. Effect of first robot importation on the number of large plant openings

Notes: The first graph in Figure 7 illustrates the number of firms in the manufacturing sector (SCIAN codes 31-33) observed in the 2018 census by opening year. The sample is further divided into four groups based on two criteria: firm size (distinguishing between firms with more or fewer than 250 workers) and the extent of robot adoption (classified as high or low robot adopters, as determined by IFR data and described in the notes for Figure 5). The second graph displays the estimated OLS coefficient derived from an event-study specification utilizing a TWFE model (equation 2). The output is the number of large manufacturing firms that opened in period *t* across CZs. The clustering of standard errors is at the CZ level. The event is defined as a first-time industrial robot importation. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data. Mexican economic census 2018.

FIGURE 8: EQUILIBRIUM IN THE LABOR MARKET



Notes: In Figure 8, I visually represent the equilibrium in the labor market and how it depends on given parameters. In equilibrium, the elasticity of labor supply (β) influences the slope of the labor supply curve, affecting the magnitude of employment and wage changes while not altering their direction. When β is relatively low (indicating inelastic labor supply), shifts in labor demand have a muted effect on employment and a heightened impact on wages. The direction of these effects is determined by the parameter σ representing the price elasticity of aggregate demand for manufacturing products. A higher σ results in a rightward shift of the labor demand curve, positively affecting employment and wages.

	C	Zs import	ting rob	ots	CZs	not impo	orting ro	bots
	A	ll CZ	Excl 2006	uding cohort	Bala sar	anced nple	N bala	ot nced
	Mean	s.d.	Mean	s.d.	Mean	s.d.	Mean	s.d.
Number of CZs	104		71		924		1612	
Number of CZs with manufacturing	104		71		730		733	
Number of CZs with manufacturing	82		52		134		134	
IMSS data (levels in 2005)								
Formal workers	1094	(6825)	423	(1468)	65	(509)	65	(508)
Monthly real wage	5350	(3950)	4739	(3697)	3487	(2560)	3487	(2558)
Formal workers in manuf.	879	(3526)	250	(791)	53	(237)	53	(237)
Monthly real wage in manuf.	5191	(3350)	4383	(3041)	3031	(1753)	3031	(1753)
Formal workers in automotive	2059	(4829)	473	(886)	145	(413)	145	(413)
Monthly real wage in automotive	6958	(4273)	5321	(3621)	3130	(1624)	3130	(1624)
Census 2000								
Population (thou.)	591.9	(2004.6)	255.6	(258.1)	33.5	(54.6)	21.9	(43.8)
Working age population (thou.)	373.6	(1307.9)	157.3	(163.4)	19.1	(33.1)	12.4	(26.4)
Share urban	76.9	(18.9)	72.3	(20.2)	42.9	(28.5)	29.7	(31.2)
Share women	51.1	(1.3)	51.2	(1.3)	51.0	(1.8)	51.1	(2.0)
Share manufacturing industry	22.3	(10.6)	19.8	(10.8)	12.1	(8.5)	10.5	(9.4)
Share automotive industry	1.9	(3.1)	1.3	(2.8)	0.3	(1.6)	0.2	(1.3)
Share agriculture industry	14.5	(11.6)	17.6	(12.2)	41.7	(17.3)	51.4	(21.0)
Share service industry	43.3	(9.9)	42.4	(11.2)	29.0	(10.6)	22.8	(12.2)
Share informal workers	26.6	(6.1)	28.1	(6.0)	43.0	(14.7)	52.9	(20.2)
Number in manuf. occupation	12.2	(3.4)	11.9	(3.5)	10.1	(6.6)	9.2	(8.2)
Share secondary complete	14.5	(4.7)	13.5	(5.0)	7.5	(4.1)	5.7	(4.2)
Share superior ed. complete	6.6	(3.2)	6.1	(3.4)	2.8	(2.1)	2.1	(2.0)
Percentage change censuses 1990–2	000 (%)							
Population	24.2	(15.9)	21.3	(16.3)	8.7	(15.6)	6.9	(16.8)
Working age population	31.6	(16.9)	29.2	(18.0)	15.9	(17.4)	12.9	(18.8)
Share urban	5.4	(12.1)	6.8	(13.8)	5.8	(22.1)	3.1	(19.7)
Share women	0.5	(1.6)	0.8	(1.5)	1.0	(3.4)	1.3	(4.7)
Share manufacturing industry	18.7	(48.7)	23.3	(56.8)	42.9	(112.8)	57.2	(161.5)
Share automotive industry	189.6	(429.6)	215.0	(499.3)	39.9	(497.1)	20.6	(364.2)
Share agriculture industry	-32.6	(23.2)	-31.8	(22.3)	-24.0	(17.3)	-18.8	(29.3)
Share service industry	14.6	(17.1)	18.6	(18.1)	68.4	(71.4)	102.8	(190.6)
Number in manuf. occupation	73.2	(52.6)	77.6	(60.9)	140.6	(233.7)	158.4	(324.7)
Share secondary complete	46.0	(28.1)	50.5	(31.2)	104.1	(118.8)	114.5	(156.7)
Share superior ed. complete	72.5	(54.8)	80.1	(62.8)	141.0	(208.3)	117.6	(203.7)

TABLE 1: COMPARISON OF CZS WITH AND WITHOUT ROBOT IMPORTS

Notes: Table 1 provides descriptive statistics comparing CZs that received robot imports during 2006–2022 (treated) to those that did not (never treated). For the treated sample, the table includes sample averages and standard deviations for all CZs. It also presents statistics for a subsample obtained by excluding the always-treated CZs, i.e., the 2006 cohort. The table also shows results for the never-treated sample for balanced and unbalanced calendar time samples. Average real monthly wage for formal workers is measured in constant 2010 pesos (adjusted using Mexican CPI). *Data sources*: IMMS public data. Mexican customs records from Secretaría de Economía. Mexican censuses 1990 and 2000.

TABLE 2: Summary statistics of import events by event cohort in $\ensuremath{\mathsf{CZs}}$

Cohort	# CZs	Mean	SD	Min	Max
2006	33	13.9	5.3	1	17
2007	13	8.4	3.9	1	15
2008	6	7.3	5.3	2	14
2009	2	3.5	0.7	3	4
2010	5	5.0	2.9	1	9
2011	6	3.8	3.2	1	10
2012	5	4.0	3.1	1	8
2013	4	2.5	1.9	1	5
2014	3	2.7	1.5	1	4
2015	5	3.8	1.6	1	5
2016	6	3.0	1.4	1	5
2018	7	2.0	1.2	1	4
2019	3	1.3	0.6	1	2
2020	2	1.0	0.0	1	1
2021	2	1.0	0.0	1	1
2022	2	1.0	0.0	1	1

Notes: Table 2 provides summary statistics for robot import events within each treated cohort available in the data. For instance, six different CZs began receiving industrial robot imports in 2011 (2011 cohort). On average, these CZs received imports for approximately 3.8 years, considering CZs that received imports only once (minimum) or for up to ten years (maximum). *Data sources*: Mexican customs records from Secretaría de Economía.

	Pre-event	Postevent								
	(1)	(2)								
A. Log Number fo	ormal workers									
1-year window	0.002	0.010								
	(0.008)	(0.007)								
2-year window	0.009	0.011								
	(0.006)	(0.008)								
3-year window	0.007	0.006								
	(0.005)	(0.014)								
4-year window	0.007	0.011								
	(0.006)	(0.023)								
B. Log average wa	(0.006) (0.023) B. Log average wage – formal workers									
1-year window	0.001	0.004								
	(0.003)	(0.003)								
2-year window	-0.000	0.004								
	(0.002)	(0.003)								
3-year window	-0.001	0.007								
	(0.002)	(0.005)								
4-year window	-0.001	0.006								
	(0.002)	(0.007)								

TABLE 3: RULING OUT COMPOSITIONAL EFFECTS

Notes: Table 3 shows the simple average of the estimated $\theta(e)$ event-study coefficients using equation 2 for different periods. The samples are restricted to cohorts included in samples balanced in event-time for different window lengths and that are observed during those periods. The event is defined as a first-time industrial robot importation. Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

	Pre-event	Postevent short run	Postevent long run
	(1)	(2)	(3)
A. Log Number formal workers			
Baseline	0.005	0.016	0.069***
	(0.004)	(0.012)	(0.029)
Balanced sample	0.005	0.005	0.058**
	(0.004)	(0.011)	(0.029)
Manufacturing	0.009	0.012	0.073
	(0.008)	(0.018)	(0.049)
Manufacturing robot adopters	0.011	0.017	0.103*
	(0.009)	(0.023)	(0.059)
Never treated	0.012***	0.041***	0.108***
	(0.004)	(0.010)	(0.021)
B. Log Average wage – formal workers			
Baseline	0.001	0.007	0.030**
	(0.002)	(0.004)	(0.013)
Balanced sample	0.002	0.005	0.027**
	(0.002)	(0.004)	(0.013)
Manufacturing	0.003	0.000	0.024
-	(0.002)	(0.008)	(0.020)
Manufacturing robot adopters	0.001	0.007	0.040*
	(0.003)	(0.010)	(0.024)
Never treated	0.001	0.006	0.026***
	(0.001)	(0.004)	(0.008)

TABLE 4: MAIN ESTIMATES FOR DIFFERENT SAMPLES

Notes: Table 4 shows the simple average of the estimated $\theta(e)$ event-study coefficients using equation 2 for different periods. Pre-event: years -5 to -1; postevent short run: years 0 to 4; long run: years 5 to 8. The outcome variable is, in turn, log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panel (b)). The event is defined as a first-time industrial robot importation. The baseline sample includes not-yet-but-eventually-treated CZs for 2005-2019, including cohorts that started importing robots during 2007–2022. The manufacturing sample is a subset of the baseline sample, focusing only on CZs within the manufacturing sector (3-digit SCIAN codes ranging from 311 to 339). The robot-adopter sample further narrows the selection to manufacturing sectors that adopt robots in the IFR dataset. Finally, the never-treated sample is an extension of the baseline sample encompassing CZs that did not receive robot imports during the analyzed period, which serves as a comparison group. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. Data sources: Mexican customs records from Secretaría de Economía. IMMS public data. 50

	Log Nur	nber Formal	Workers	Log	g Average W	age
	Pre-event	Postevent short run	Postevent long run	Pre-event	Postevent short run	Postevent long run
	(1)	(2)	(3)	(4)	(5)	(6)
A. By age group						
Aged 15–19	0.006	0.004	0.032	-0.000	-0.004	0.001
	(0.007)	(0.027)	(0.044)	(0.002)	(0.007)	(0.012)
Aged 20-34	0.006	0.011	0.071**	-0.000	0.004	0.029***
	(0.004)	(0.012)	(0.035)	(0.002)	(0.004)	(0.012)
Aged 35–49	0.006*	0.001	0.028	0.000	0.007*	0.026**
	(0.003)	(0.012)	(0.026)	(0.002)	(0.004)	(0.013)
Aged 50-64	0.007***	0.016	0.085***	0.005***	-0.007	0.008
	(0.003)	(0.011)	(0.023)	(0.002)	(0.007)	(0.015)
Age >64	0.009*	0.006	0.007	0.007	-0.003	0.029
	(0.005)	(0.015)	(0.040)	(0.004)	(0.010)	(0.027)
B. By wage group						
Low-wage group	0.002	0.003	0.035	0.002*	-0.002	0.009
	(0.004)	(0.011)	(0.031)	(0.001)	(0.004)	(0.011)
Middle-wage group	0.002	0.019	0.070*	0.001	0.002	0.007
	(0.005)	(0.017)	(0.037)	(0.001)	(0.002)	(0.005)
High-wage group	0.003	-0.006	0.024	-0.002*	0.001	0.002
	(0.006)	(0.018)	(0.036)	(0.001)	(0.003)	(0.009)
C. By firm Size						
1–5 workers	0.006*	0.005	0.037	0.002	0.003	0.027**
	(0.003)	(0.011)	(0.024)	(0.002)	(0.006)	(0.013)
6–50 workers	-0.003	-0.003	0.023	0.001	-0.003	0.017
	(0.003)	(0.011)	(0.021)	(0.001)	(0.005)	(0.012)
51–250 workers	0.007	0.004	0.041	-0.003	-0.003	-0.006
	(0.005)	(0.018)	(0.030)	(0.003)	(0.007)	(0.022)
>250 workers	-0.001	0.028*	0.023	0.002	-0.007	0.014
	(0.005)	(0.015)	(0.038)	(0.003)	(0.008)	(0.020)
D. By CZ size						
Excluding >500k hab.	0.004	0.022	0.096***	0.002	0.012**	0.038***
-	(0.003)	(0.014)	(0.034)	(0.002)	(0.005)	(0.016)
Excluding >150k hab.	0.004	0.017	0.133**	0.002	0.023***	0.049***
	(0.005)	(0.022)	(0.057)	(0.002)	(0.008)	(0.020)

TABLE 5: HETEROGENEITIES

Notes: See notes in Table 4. Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

	Pre-event	Postevent short run	Postevent long run
	(1)	(2)	(3)
A. Log Number formal workers			
Baseline CS	0.005	0.016	0.069***
	(0.004)	(0.012)	(0.029)
Import value event	< 0.001	0.028	0.383***
	(0.007)	(0.025)	(0.071)
Baseline DD	-0.003	0.033*	0.100***
	(0.011)	(0.019)	(0.043)
DD with intensity	-0.001	0.031*	0.066***
	(0.016)	(0.016)	(0.026)
B. Log Average wage – formal workers			
Baseline CS	0.001	0.007	0.030**
	(0.002)	(0.004)	(0.013)
Import value event	-0.001	-0.001	0.067**
	(0.002)	(0.010)	(0.030)
Baseline DD	-0.007	0.017	0.046
	(0.008)	(0.013)	(0.030)
DD with intensity	-0.014**	0.017***	0.035***
	(0.007)	(0.006)	(0.010)

TABLE 6: OTHER ROBOT ADOPTION MEASURES

Notes: Table 6 shows the event-study estimates using different robot adoption measures. *Baseline CS* corresponds to the main specification in Table 4. *Import value event* refers to the same model but defines the event as the period when the cumulative total value of robot imports is larger than 50% of the CZ's 2005 total wage bill (refer to Figure A15 to see estimates under different thresholds). *Baseline DD* refers to the estimation of a TWFE model defining the event as first-time receipt of industrial robot imports. *DD with intensity* refers to *Baseline DD* interacting the event with a measure of robot importation intensity, the cumulative number of years receiving robot imports. Pre-event: years -5 to -1; postevent short run: years 0 to 4; long run: years 5 to 8. The outcome variable is, in turn, log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panel (b)). Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

	Mean	Pre-event	Event year	Postevent short run	Postevent long run
	(1)	(2)	(3)	(4)	(5)
A. Formal employment and earnings					
Log number workers (IMSS)	54122.5	0.014***	0.003	0.044**	0.118
		(0.006)	(0.008)	(0.020)	(0.076)
Log number workers (ENOE)	46452.3	0.042	-0.054	0.036	0.300
		(0.035)	(0.072)	(0.086)	(0.224)
Log average wage (IMSS)	208.6	< 0.001	0.008*	0.008	0.012
		(0.003)	(0.004)	(0.008)	(0.016)
Log hours worked	49.4	< 0.001	0.016	-0.013	-0.031*
		(0.003)	(0.017)	(0.009)	(0.018)
B. Informal employment					
Log number informal workers	84293.8	0.033*	-0.139**	-0.043	0.169*
		(0.018)	(0.061)	(0.070)	(0.099)
Informality rate	60.2	0.163	-2.111**	-1.016	1.600
		(0.451)	(1.043)	(1.423)	(2.068)
C. Unemployment					
Log number unemployed	6235.5	0.018	-0.497	-0.203	0.313
		(0.095)	(0.308)	(0.246)	(0.329)
Unemployment rate	0.04	-0.001	-0.003	0.000	-0.002
		(0.001)	(0.004)	(0.004)	(0.005)
D. Educational attainment					
Log number less than primary	27473.5	-0.030	0.063	0.054	0.199
		(0.039)	(0.200)	(0.260)	(0.248)
Log number primary complete	121210.9	0.021	-0.082	-0.025	0.094
		(0.018)	(0.064)	(0.074)	(0.108)
Log number secondary complete	51878.8	0.069	-0.067	-0.062	0.209
		(0.053)	(0.064)	(0.093)	(0.142)
Log number tertiary complete	30868.6	0.121	0.018	0.056	0.813*
		(0.076)	(0.197)	(0.218)	(0.455)
E. Other					
Log number working-age pop.	231568.3	0.020	-0.079	-0.005	0.158
		(0.017)	(0.056)	(0.065)	(0.099)
Log number pop. in LFP	149827	0.027	-0.102*	-0.026	0.128
		(0.017)	(0.055)	(0.065)	(0.095)
Log number living in urban area	134449.7	0.001	-0.002	0.011*	0.034*
		(0.002)	(0.003)	(0.006)	(0.020)

TABLE 7: ADDITIONAL LABOR MARKET OUTCOMES FROM ENOE DATA

Notes: See notes in 4. Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. Sample is restricted to CZs available in ENOE. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data. ENOE data.

BETWEEN 2011 ANI	0 2015			TAIN MODE		COMES	1163 CJCZ	DNIED CETEM				
A: Population Census	es outcome	es. Long diffe	rences 2010	-2015								
	All indust	tries										
	Employm	ient-to-pop	Wage-emj	pl-to-pop	Self-emp	l-to-pop	Labor For	ce Particip. rate	Unemplo	yment rate	Share In	formal
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
MX exposure to robots 2011-2015	-0.4 (1.6)	-0.4 (2.1)	-0.5 (1.4)	-0.5 (1.6)	0.1 (1.1)	0.1 (1.2)	-0.3 (1.4)	-0.4 (1.9)	-0.7 (1.1)	-0.7 (1.2)	0.4 (1.7)	0.4 (1.6)
Observations Adj. R-squared	1,804 0.154 45.47	19	1,804 0.0531 23 50	19	1,804 0.180 21.07	19	1,804 0.137 50 30	19	1,798 0.0231 4.406	19	1,797 0.155 40 12	19
Lower limit CI Lower limit CI	-3.452 2 703	-4.502 3 713	-3.211	-3.571 2 565	-1.974	-2.156 2.373	-3.165 2.479	-4.099 3.388	-2.803	-3.056 1 659	-2.86	-2.793 3.606
First-stage Coefficient First-stage F-statistic	0.26 4.895		0.26 4.895	i	 0.26 4.895		0.26 4.895		0.266 5.169		0.266	
B: IMSS data - Insured	ł workers.	Long differe	10-2010-20	115								
	All indust	tries					Manufactı	uring industry				
	Ln Numb	er Workers	Ln Pa	yroll	Ln Mea	n Wage	Ln Nun	uber Workers	LnF	ayroll	Ln Mear	ו Wage
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
MX exposure to robots 2011-2015	5.0 (10.9)	5.1 (9.7)	4.2 (12.5)	4.2 (8.6)	-0.9 (4.5)	-0.8 (2.6)	-26.3 (28.4)	-26.1 (16.1)	-19.7 (27.7)	-19.6 (15.5)	6.5 (10.1)	6.5 (6.9)
Observations Adj. R-squared Mean dep. 2005	1,211 0.010 12050	19	1,211 0.014 3004000	19	1,211 0.008 158.10	19	777 0.067 3485	19	777 0.037 954484	19	777 0.031 134.20	19
Lower limit CI Upper limit CI	-16.44 26.5	-13.98 24.12	-20.29 28.64	-12.57 21.02	-9.682 7.97	-5.979 4.29	-81.98 29.47	-57.54 5.344	-74.04 34.54	-49.96 10.83	-13.22 26.23	-7.108 20.17
First-stage coefficient First-stage F-statistic	0.252 3.976		0.252 3.976		0.252 3.976		0.24 3.043		0.24 3.043		0.24 3.043	
<i>Notes</i> : This table presents 2 ² and Panel (b) those based on by CZ share in national worl characteristics of CZs in 199(employment in manufacturi and the second column pres 2000 censuses. IFR data. Fab	LS estimates the IMSS dat uing-age populat 0 (log populat ng, services a ents robust st er (2020).	: of the effects of aset. In all mode ulation in 1990. J tion; share urban ind agriculture s andard errors co	exposure to re ls, I instrumer The covariates ; share womer ectors). For ea mputed follow	bots on chang tt Mexican exp included in ea n; share popul ch outcome, ti ving Borusyak	ges in labor n oosure to robo och model are ation over 65 he first colun c et al. (2019)	narket outcor ots using expo e region dum ; share popul on presents s in parenthes	mes between 20 soure to robots mies, exposure lation with high tandard error ϵ es. *** $p < 0.01$,	110 and 2015. Panel from European court to US robots (2011–; to US robots (2011–; a school complete; slimates that are rob stimates that are rob $^{**} p < 0.05, ^* p < 0.15$	 a) presents ou tries. All IV es tries. Ol 5), exposure are populatio ust to heteros ust to heteros 	ttcomes based o stimates are from e to maquiladoras n with college o kedasticity and IMMS public d	n population 1 regressions 3 in 1990, der omplete; and correlation w ata. Mexican	censuses weighted nographic shares of rithin CZ, 1990 and

	Number of firms	Labor share	Gross value added	GVA per worker	Total fixed assets	Capital per worker	Input cost	Total payroll	Number workers	Workers prod. and sales	Workers admin	Hours worked	Hours prod. and sales	Hours admin
			(log)	(log)	(log)		(log)	(log)	(log)	(log)	(log)	(log)	(log)	(log)
	(1)	(2)	(3)	(4)	(2)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
Panel (a). All i D_{mt}^{post}	ndustries 0.012 (0.013)	-0.011 (0.068)	0.054^{*} (0.030)	0.028 (0.024)	0.017 (0.043)	-0.019 (0.014)	0.060** (0.025)	0.025 (0.029)	0.018 (0.015)	0.009 (0.023)	0.000 (0.028)	0.020 (0.015)	0.011 (0.024)	-0.006 (0.028)
Observations Adj. R-squared Mean	93,464 0.932 27.80	32,979 0.131 0.330	32,352 0.849 14.83	32,352 0.626 0.0642	32,908 0.803 19.09	43,054 -0.0818 0.0508	32,982 0.875 14.86	30,945 0.852 4.812	43,054 0.890 200.5	40,283 0.842 98.15	27,181 0.751 24.59	42,913 0.885 501.9	40,149 0.832 245.9	27,071 0.737 60.44
Panel (b). Man D_{mt}^{post}	ufacturing 0.024 (0.024)	; sector: № -0.025 (0.056)	Metal, m a 0.301* (0.156)	ichinery a 0.169* (0.098)	nd equip 0.351** (0.159)	ment, po 0.026* (0.014)	wer gen 0.298** (0.134)	ration, tr 0.304** (0.139)	ansportatic 0.072 (0.073)	m, plastic 0.099 (0.095)	s and rubb -0.040 (0.127)	er 0.077 (0.078)	0.105 (0.105)	-0.022 (0.131)
Observations Adj. R-squared Mean	5,509 0.887 12.50	769 0.0558 0.400	$761 \\ 0.794 \\ 15.78$	761 0.608 0.0521	765 0.751 11.99	983 0.190 0.0478	769 0.797 21.20	695 0.782 8.296	983 0.802 164.1	901 0.755 98.48	586 0.669 19.18	983 0.786 359.7	901 0.738 224.1	586 0.662 42.48
<i>Notes</i> : Table 9 show industries and Pant and other manufact variables are balanc value added (colum per worker (colum number of workers, Columns (12)–(14) s started importing n except for the numb and * denote statisti	<i>s</i> ^s the results el (b) for the f urring not incl re-sheet meas m (2)); log inpu (2)); log inpu (5)); log inpu (5)); log inpu (10) show similar obots durring ser of worker cal significan	of estimatii ollowing n luded in Ta ures captu u per work for the log for the log results for 2007–2022 s, number ce at the 1°	ng the TWI nanufactur ble 10. The ring firm p ter, calculat olumn (6)); r, number of the log nur the log nur the and nours w 5%, and	E model, ec ing industri first depent roductivity ed as the to log total wa log total wa i workers in mber of hou re calculatec orked in a y 10% levels,	quation 3, w es: basic m es: basic m and size, ag tal fixed as ge bill (coli production production rs worked A for the pe rear (in tho respectivel	where the event ether and the event of the event of the log streads the log stread the event of the log sets divided the sets divided the nume (7)); and sales, in a year. The event of the evento evento event of the event of th	ent is defir l products, number of number of the CZ -fit a by the nu d log fixed and colum he sample e the event, d shares (r ccs: Mexico	the as a first machinery, firms at the we-digit sec we-digit sec mber of wor a lassets (as a n (11) for the includes no includes no in levels, a atios). Standan economic	-time industr equipment, e CZ-five-digit tor level: labc or level: labc arkers (colum 1 proxy for ca te log numbe: t-yet-but-eve: und are repor ard errors ar ard errors ar censuses 200.	ial robot imp electronics, a sector level rr share, calc n (3)); log to pital, columu r of workers ntually-treat ted in millio e clustered <i>a</i> 33, 2008, 2013	oortation. Pau ppliances, pc (column (1)). ulated as the tal gross valu n (8)). Colum in administr: ed CZs for 2(ms of Mexica th the CZ leve 3, and 2018.	nel (a) preser wer generat The followi total wage b ae added (cc m (9) presen 05–2019, inc m pesos (CF and are in	this results f tion, transp ng seven de nill divided olumn (4)) <i>i</i> ts results fc coluding mu cluding coh l'adjusted parenthese	or all the ortation, pendent by gross and GVA transformed to the GVA transformed to the log unagers). orts that to 2005), s. ***, **,

TABLE 9: MECHANISMS

	Number of firms	Labor share	Gross value added	GVA per worker	Total fixed assets	Capital per worker	Input cost	Total payroll	Number workers	Workers prod. and sales	Workers admin	Hours worked	Hours prod. and sales	Hours admin
			(log)	(log)	(log)		(log)	(log)	(log)	(log)	(log)	(log)	(log)	(log)
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
Panel (a). Ma D_{mt}^{post}	nufacturii -0.014 (0.023)	ng: Wood -0.018 (0.031)	and pap 0.031 (0.111)	er, chemi 0.014 (0.066)	cal indus 0.063 (0.116)	s tries 0.032** (0.013)	-0.013 (0.100)	-0.026 (0.114)	-0.084 (0.066)	-0.203** (0.100)	-0.195* (0.111)	-0.073 (0.067)	-0.203** (0.101)	-0.185 (0.114)
Observations R-squared Mean	5,356 0.923 24.10	1,068 0.393 0.346	1,036 0.799 32.84	1,036 0.612 0.0933	1,063 0.790 32.55	1,433 0.581 0.0542	1,068 0.819 74.43	1,014 0.754 6.827	1,433 0.843 285.4	1,354 0.753 140.3	929 0.649 30.88	1,433 0.842 731.1	1,354 0.755 382.7	929 0.632 77.46
Panel (b). Mé D ^{post}	nufacturi -0.028 (0.030)	ng: Textil 0.019 (0.054)	es, appar 0.042 (0.093)	el, leathe 0.051 (0.060)	r goods , -0.009 (0.118)	food, and 0.005 (0.010)	beverage 0.030 (0.081)	e industri 0.144 (0.113)	es -0.094 (0.063)	-0.017 (0.091)	-0.072 (0.097)	-0.070 (0.065)	0.003 (0.096)	-0.065 (0.099)
Observations R-squared Mean	7,393 0.888 15.32	1,721 0.0699 0.388	1,715 0.751 15.30	$\begin{array}{c} 1,715\\ 0.516\\ 0.0419\end{array}$	1,690 0.729 15.72	2,235 0.521 0.0378	1,721 0.787 27.63	1,443 0.739 7.844	2,235 0.783 227.6	1,902 0.739 177.7	969 0.727 33.62	2,235 0.773 496.6	1,902 0.726 393.5	969 0.715 75.50
Panel (c). No D_{mt}^{post}	nmanufac 0.017 (0.013)	t uring -0.012 (0.075)	0.050* (0.030)	0.025 (0.024)	0.009 (0.043)	-0.023 (0.016)	0.059** (0.025)	0.015 (0.027)	0.027* (0.014)	0.016 (0.022)	0.011 (0.029)	0.027* (0.015)	0.017 (0.023)	0.003 (0.029)
Observations R-squared Mean	75,206 0.935 30.28	29,421 -0.0462 0.325	28,840 0.856 14.11	28,840 0.629 0.0648	29,390 0.807 18.96	38,403 -0.0718 0.0514	29,424 0.881 11.72	27,793 0.861 4.495	38,403 0.899 196.5	36,126 0.852 92.28	24,697 0.757 24.11	38,262 0.894 496.7	35,992 0.841 233.2	24,587 0.742 59.58
<i>Notes</i> : Table 10 sl following manufi textile, clothing, J the log number o aggregated to the assets divided by bill (column (7)); in production ank of hours worked calculated for the in a year (in thou levels, respectivel	nows the resturing indu- eather, word, f firms at though the CZ-five-dig - CZ-five-dig - the number and log five - and log five - period befor period befor sands), and sourc	ults of estin- ustries: fooc d, paper, pr e CZ-five-d git sector lev id assets (as column (11) he sample i re the event, shares (ratio 22: Mexican	atting the T atting the T diriting, and ligit sector (e: labor sh s (column b a proxy fo ncludes no in levels, a s). Standar economic	WFE mode bacco, ener l nonmetalli level (colun are, calcula (3)); log totl (3)); log totl (3)); log totl ar capital, cc g number o t-yet-but-ev nd are repo d errors are censuses 20	L, equation igy, fossil fri gy, fossil fri mn (1)). Th mn (1). Th ted as the al gross ve al gross ve ilumn (8). f workers. rentually-t- rrted in mil 3, 2008, 20	n 3, where the uels, chemic uels, chemic uels, chemic products. F he following to tal wage l to tal wage l alue added Column (5 in administ reated CZs lions of Me) at the CZ le 013, and 201	the event is c als, plastics anel (c) into is even dep ill divided (column (4 (column (4)) presents 1 (presents 1) presents 1 (for 2005–20 for 2005–20 for 2005–21 for 2005–20 for 2005–2005–20 for 2005–20 for 2005–20 for 2005–2005–2005 for 2005–2005–2005 for 2005–2005–2005 for 2005–2005 for 200	lefined as a , and rubbe fludes result endent vari by gross ve () and GVA results for t results for t (including 1 119, including 1 (CPI-adjust CPI-adjust	first-time ind r: Panel (b) p ts for all non ables are balk ables are balk ables are vorker per worker he log numb managers). C ng cohorts th re dt o 2005), e eses. ***, **, a	ustrial robol resents resu manufacturi manufacturi molumn (2) ; er of worker er of worker at started in xcept for the md * denote	t importation. Its for the foll ing industries neasures captita apital per wor (14) iog input co -(14) show si mporting robc robc inter of w statistical sig	Panel (a) pi owing manu owing manu orker, calcula sets (columu sets (columu)) for the log milar result milar result tets during 2(orkers, num nificance at	esents result facturing in the deturing in the deturing in the deturbance of the deturbance of the the the deturbance of the the the deturbance of the	Its for the industries: ariable is and size, otal fixed tral wage f workers g number deans are s worked and 10%

Appendices

A	Additional Figures	58
B	Additional Tables	74

Appendix A Additional Figures



FIGURE A1: TRENDS IN THE MEXICAN FORMAL SECTOR







Notes: Figure A1 illustrates the trends in the number of formal workers (measured in millions) and average monthly real wages (adjusted to 2010 Mexican pesos) over time. In Panel (a), the sample of formal workers is divided into three distinct wage categories: those earning two minimum wages or less (low wage), those earning between two and five minimum wages (medium wage), and those earning more than five minimum wages (high wage). In Panel (b), the sample is divided into two groups: CZs receiving industrial robot imports at any point during the analysis period and CZs not receiving such imports. *Data sources*: IMMS public data, Mexican customs records from Secretaría de Economía.

FIGURE A2: CHARACTERIZATION OF MEXICAN MANUFACTURING SECTOR



Monthly real wage in terms of the real Minimum Wage

(a) Fraction of workers for each wage bin



(b) Fraction of wage group for each occupation

Notes: Panel (a) of Figure A2 shows the histogram of the log average monthly real wages (adjusted to 2005 Mexican pesos) in terms of the monthly minimum wage for either all workers or formal workers. The vertical lines divide formal workers into three distinct wage categories: those earning two minimum wages or less (low wage), those earning between two and five minimum wages (medium wage), and those earning more than five minimum wages (high wage). Panel (b) shows for each occupation the fraction of formal workers in each wage group. For example, 73% of industrial operators fall into the medium-wage group. *Data sources*: Own elaboration based on ENOE 2005, 4th quarter.

FIGURE A3: CHARACTERIZATION OF MEXICAN MANUFACTURING SECTOR



(c) Average monthly wage by age

Notes: Figure A3 illustrates the average wage trends for different groups of workers across various age brackets. The vertical axis of the figure represents the average monthly wages, measured in terms of the monthly minimum wage, for three distinct groups: all workers (across all industries), all manufacturing workers, and all formal manufacturing workers (those registered in IMSS). *Data sources*: Own elaboration based on ENOE 2005, 4th quarter.

FIGURE A4: INDUSTRIAL ROBOTS IN MEXICO: LARGE INCREASE BUT EARLY STAGE



(B) Automation by industry

Notes: Panel (a) of Figure **??** presents a comparison of the cumulative stock of industrial robots in Mexico with that in other developed countries, specifically the United States (US) and European Union (EU) countries. It is important to note that IFR data for Mexico and the US are available only for years from 2011. To create the trend data before 2011, I impute values using the aggregate stock for North America and distributed among Mexico and the US based on their relative sizes in 2011. Panel (b) of the figure depicts the same cumulative robot stock for Mexico, but it further breaks down this stock for different 3-digit industries within the manufacturing sector. *Data source*: IFR.

Figure A5: Number of CZs that started receiving robot imports by year





FIGURE A6: EVOLUTION OF NUMBER OF FORMAL WORKERS BY TREAT-MENT COHORT



(b) Including every CZ (in gray)

Notes: Figure A6 shows the raw means for the number of formal workers by treatment cohort over event-time. Estimates are relative to the outcomes one year before the event (period -1). Panel (a) shows cohort averages across CZs. Panel (b) shows the same figures as in Panel (a) and overlaps the raw means for each CZ separately in gray. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.





Notes: Figure A7 plots the estimated $\theta(e)$ event-study coefficients using equation 2 for three wage categories of formal workers: those earning two minimum wages or less (low wage), those earning between two and five minimum wages (medium wage), and those earning more than five minimum wages (high wage). The outcome variable is log number of formal workers. The event is defined as a first-time industrial robot importation. The vertical lines reflect the 95% uniform confidence intervals, which are robust to multiple hypothesis testing. Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

FIGURE A8: HETEROGENEITY IN DYNAMIC EFFECTS ON FORMAL EMPLOY-MENT AND WAGES

.15

-

Aged 15-19

Aged 20-34

Aged 35-49



.... Aged 50-64 Aged >64 .05 - 05 -8 -7 -6 -5 -3 -2 -1 Ó 2 5 6 Periods since the event

(a) Ln formal workers by **age group**



(c) Ln formal workers by **firm size**



(e) Ln formal workers by CZ size











Notes: Figure A8 plots the estimated $\theta(e)$ event-study coefficients using equation 2. The outcome variable is, in turn, the log number of formal workers (Panels (a), (c), and (e)) and the log average real monthly wage for formal workers, deflated to constant pesos using Mexican CPI (Panels (b), (d), and (f)). The event is defined as a first-time industrial robot importation. The vertical lines reflect the 95% uniform confidence intervals, which are robust to multiple hypothesis testing. Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

FIGURE A9: ALTERNATIVE MEASURES OF ROBOT ADOPTION EVENT



(a) Histogram of imputed and not-imputed values of robot imports



(b) Histogram of original event and alternative events based on import value

Notes: See main text for further explanation. Data sources: Mexican customs records from Secretaría de Economía. IMMS public data.







Notes: Figure A10 overlaps the estimated event-study coefficients using three different methods: CS (Callaway and Sant'Anna (2021)), SA (Sun and Abraham (2021)), and DD (two-way fixed effect model). Estimates are relative to the outcomes one year before the event (period -1). The outcome variable is, in turn, the log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panel (b)). The event is defined as a first-time industrial robot importation. The shaded area and vertical lines reflect the 95% pointwise confidence intervals for DD and SA and uniform confidence intervals for CS. The clustering of standard errors is at the CZ level. Sample of not-yet-but-eventually-treated CZs for 2005–2019, including cohorts that started importing robots during 2007–2022. The unit of observation is the CZ–4-digit industry (N=100,619; 71 CZs). *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.



Figure A11: Pretrends sensitivity analysis

Notes: Figure A11 shows the result of a sensitivity analysis proposed by Rambachan and Roth (2023) over the coefficients estimated under the CS method. This analysis estimates the post-treatment effect under different assumptions about parallel trend violations, demonstrating the extent to which the postevent trend differences would need to deviate from pretrends to render the estimates negative or statistically insignificant. It provides confidence intervals that enable the post-treatment departure from parallel trends to reach a magnitude of up to *Mbar* times the maximum pretreatment deviation, with *M* being a variable that can take different values. The outcome variable is, in turn, the log number of formal workers (Panels (a) and (b)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panels (c) and (d)). The event is defined as a first-time industrial robot importation. Short term refers to the year of the event, and long term refers to six years after the event occurs. The clustering of standard errors is at the CZ level. Sample of not-yet-but-eventually-treated CZs for 2005–2019, including cohorts that started importing robots during 2007–2022. The unit of observation is the CZ–4-digit industry. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

FIGURE A12: TWFE MODELS: DD & DDD



Notes: Figure 5 displays the estimated event-study coefficients obtained using equation 2, a TWFE model. The unit of observation is the CZ–3-digit industry. In Panel (a), the results of the difference-in-differences specification are presented, showing overlapping estimates from two different subsamples: "high robot adopters", comprising manufacturing industries with higher rates of robot adoption observed in the IFR data, and "low robot adopters", which include the remaining nonmanufacturing industries. Panel (b) shows estimates from the triple difference-in-differences specification, where the third difference is introduced by contrasting high and low robot adopters. The outcome variable is, in turn, the log number of formal workers and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI. The event is defined as a first-time industrial robot importation. The clustering of standard errors is at the CZ level. Sample of CZs with population smaller than 500 thousands inhabitants for the period 2005–2019, including cohorts that started importing robots during 2007–2022. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.



Figure A13: SDD estimates for CZ #24050 (2011 cohort)

Notes: Figure A13 exhibits the estimated treatment effects using the synthetic difference-in-difference model as outlined by Arkhangelsky et al. (2021) for a specific CZ (belonging to 2011 cohort) that experiences a statistically significant positive formal employment and wage effect. The comparison group for these estimates is derived from the pool of all CZs that have never been treated, and it employs optimal weighting of units and pre-event periods in its construction. The dataset comprises a panel of CZs, balanced in calendar time, spanning from 2005 to 2019. The outcome variables are the logarithm of the number of formal workers and the logarithm of the average real monthly wage for formal workers, adjusted to constant 2010 pesos using Mexican CP. The event is defined as a first-time industrial robot importation. Panels (a) and (c) present the raw means of the treated CZ and the synthetic control. Panels (b) and (d) show the corresponding difference between treatment and control over event-time. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

FIGURE A14: COMPARISON OF 2SLS ESTIMATES WITH RESULTS FROM ACEMOGLU AND RESTREPO (2020)



Figure A14 compares OLS and IV estimates from Acemoglu and Restrepo (2020) for the US and my own replication for Mexico. BJH refers to the standard error estimator proposed by Borusyak et al. (2021). *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data. Mexican censuses 1990 and 200. IFR data. Faber (2020).
FIGURE A15: ALTERNATIVE EVENT DEFINITION USING ROBOT IMPORTS VALUE



(a) Ln Number of formal workers

(b) Ln Avg. monthly wage – formal workers

Notes: Figure A15 plots the simple average of estimated $\theta(e)$ event-study coefficients using equation 2 (Callaway and Sant'Anna (2021) method) for different event-periods. The event is defined as the period where the cumulative total value of robot imports is larger than X% of the CZ's 2005 total wage bill (where X% varies on the horizontal axes). Pre-event: years -5 to -1; postevent short run: years 0 to 4; long-run: years 5 to 8. The outcome variable is, in turn, log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panel (b)). Clustering of standard errors is at the CZ level. Sample of not-yet-but-eventually-treated CZs for the period 2005–2019, including cohorts that started importing robots during 2007–2022. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

Figure A16: CONTROLLING FOR PLANT OPENINGS BY LARGE FIRMS



Notes: Figure A16 shows the estimated event-study coefficients using a two-way fixed effect model, controlling for the opening of large (more than 250 workers) manufacturing firms (which survive up to the 2018 Mexican economic census). In the first graph, the outcome variable is the log number of formal workers, and in the second graph, it is the log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI. The clustering of standard errors is at the CZ level. The event is defined as a first-time industrial robot importation. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data. Mexican economic census 2018.

Appendix B Additional Tables

TABLE B1: CHARACTERIZATION OF MEXICAN MANUFACTURING SECTOR BY WAGE GROUP

		All worker	s	Manu	facturing IM	SS workers
	Low wage	Medium wage	High wage	Low wage	Medium wage	High wage
	(1)	(2)	(3)	(4)	(5)	(6)
A. By occupation						
Professionals	3%	7%	21%	1%	4%	19%
Education workers	2%	5%	17%	0%	0%	1%
Managers	0%	1%	8%	0%	0%	10%
Office workers	7%	14%	20%	6%	8%	19%
Industrial operators	25%	35%	15%	80%	74%	37%
Merchants	15%	10%	7%	4%	5%	8%
Transport operators	3%	7%	6%	1%	4%	3%
Workers in personal services	29%	11%	2%	7%	3%	1%
Protection and surveillance work	1%	5%	3%	0%	1%	1%
Agricultural	17%	5%	1%	0%	0%	1%
Total	100%	100%	100%	100%	100%	100%
N (mill.)	7.84	13.54	3.84	0.59	2.30	0.44
B. By age group						
15–19	19%	8%	1%	13%	7%	0%
20–24	18%	17%	6%	25%	20%	7%
25–29	12%	17%	13%	15%	19%	17%
30–34	11%	15%	17%	15%	16%	20%
35–39	10%	13%	17%	11%	13%	19%
40-44	8%	11%	17%	9%	11%	16%
45–49	6%	8%	14%	5%	6%	10%
50-54	5%	5%	8%	3%	4%	6%
55–59	4%	4%	4%	2%	3%	3%
60–64	3%	1%	3%	1%	1%	1%
65+	3%	1%	1%	0%	0%	0%
Total	100%	100%	100%	100%	100%	100%
N (mill.)	7.62	13.51	3.84	0.58	2.29	0.44

Notes: Panel (a) of Table B1 shows the share of workers in each occupation within each wage group for the sample of all workers and the subsample of workers in the manufacturing sector who are registered in IMSS. Panel (b) shows similar statistics by age group. Wage categories: those earning two minimum wages or less (low wage), those earning between two and five minimum wages (medium wage), and those earning more than five minimum wages (high wage). *Data sources*: Own elaboration based on ENOE 2005, 4th quarter.

TABLE B2: Growth in Robot Adoption by industry 2015-2010 (%)

	N workers	C	hange in s	stock of re	obots 201	0-2015 (%)
	MEX 2000	MEX	US	EU9	EU7	EU5	LA
All industries	31049.6	694.8	38.2	12.7	2.9	-0.2	96.1
Automotive	460.6	662.3	108.2	3.2	-13.1	-21.6	216.9
Rubber & Plastic	138.5	711.2	92.3	62.8	50.8	45.9	89.5
Industrial Machinery	87.6	1077.1	1485.7	72.9	82.8	79.2	474.2
Electronics	248.1	4714.3	119.3	-6.3	-23.8	-27.4	70.7
Other Manufacturing	739.8	303.9	877.8	-23.2	15.9	45.8	393.3
Food & Beverages	1153.3	2300.0	97.9	49.3	69.3	76.8	344.7
Basic Metals	126.1	1455.6	10162.7	28.8	70.6	65.1	315.0
Metal Products	415.3	322.6	31.9	23.1	8.8	2.8	114.0
R+D	2594.1	489.5	369.9	2.1	22.4	43.6	418.5
Pharmaceuticals	284.3	4850.0	104.6	116.3	130.5	131.9	121.9
Electrical Machinery	204.4	136.4	11554.5	107.5	83.4	81.5	200.0
Other Services	16312.5		279.5	313.1	153.2	590.9	
Minerals	457.1		401.2	-3.4	1.2	2.5	475.0
Paper	236.4		2046.2	12.1	40.9	45.8	300.0
Utilities	159.3			55.6	30.2	152.0	
Wood Products	571.0		1136.4	-25.4	5.5	13.0	150.0
Construction	2767.1		173.9	36.0	32.1	108.7	1200.0
Mining	197.7		1750.0	-29.8	-33.3	-22.7	440.0
Textiles	1474.9		900.0	0.0	-12.1	-3.6	
Agriculture	5678.9		666.7	32.3	15.5	16.5	1600.0
Stock of robots							
in year 2000		963	87999	187480	95756	70249	1300
in year 2010		1855	169550	306760	157492	115105	6514
in year 2015		14743	234245	345770	162070	114883	12775

Notes: Table B2 shows the number of workers in Mexico 2000 and the change in the stock of robots between 2010 and 2015, calculated as the % change relative to the base year. EU9 includes Germany, Denmark, Spain, Finland, France, UK, Italy, Norway, and Sweden; EU7 excludes Germany and Norway; EU5 excludes the UK and Spain; LA includes Argentina, Brazil, Chile, Colombia, Peru and Venezuela. *Data sources*: IFR data. Mexican population census 2000.

75

TABLE B3: LABOR MARKET VARIABLES DO NOT PREDICT EVENT TIMING AFTER INCLUDING FIXED EFFECTS

		All indu	stries		Ν	lanufa	octuring	g industr	ies
	(1)	(2)	(3)	(4)	(5)		(6)	(7)	(8)
Ln Number of formal workers	0.027***	0.0904***	0.005	0.005	0.019	*** 0.	059***	0.002	0.001
	(0.007)	(0.009)	(0.004)	(0.004)	(0.00	5) (().009)	(0.005)	(0.005)
Ln Avg. monthly formal wage	0.046***	0.098***	0.008	0.004	0.0903	*** 0.	080***	0.005	0.006
	(0.013)	(0.021)	(0.012)	(0.013)	(0.02	0) (().027)	(0.014)	(0.017)
$CZ \times 4$ -dig. sector FE		x	x	x			x	x	x
Year FE			х					x	
Year \times 4-dig. sector \times Region FE				x					x
Observations	100,619	100,423	100,423	98,988	23,93	5 2	3,873	23,873	23,241
Adj. R-squared	0.022	0.334	0.660	0.655	0.02	5 ().336	0.666	0.658
P-val F-stat	0.00	0.00	0.25	0.40	0.00)	0.00	0.77	0.90

Notes: Table B3 shows the estimates of a linear probability model where the outcome is a binary variable indicating whether the CZ started receiving industrial robot imports. The model includes two explanatory variables at the CZ–4-dig. sector–year level: log number of formal workers and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI. The last row displays the p-value of a joint significance test for the two included covariates. Clustering of standard errors is at the CZ level. ***, ***, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. Refer to notes in Table 4 for the definitions of the different samples. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

	Pre-event	Postevent short run	Postevent long run
	(1)	(2)	(3)
A. Log Number formal Workers			
Baseline	0.003	0.008	0.055**
	(0.004)	(0.011)	(0.028)
Balanced sample	0.002	0.001	0.049
	(0.004)	(0.012)	(0.030)
Manufacturing	0.005	0.003	0.073
	(0.007)	(0.019)	(0.048)
Manufacturing robot adopters	0.006	0.000	0.085
	(0.008)	0.023	(0.058)
Never treated	0.011***	0.036***	0.105***
	(0.003)	(0.009)	(0.019)
B. Log Average Wage – formal workers			
Baseline	0.002	0.006	0.028**
	(0.002)	(0.004)	(0.013)
Balanced sample	0.002	0.006	0.026**
	(0.002)	(0.005)	(0.013)
Manufacturing	0.004	0.001	0.024
	(0.002)	(0.007)	(0.019)
Manufacturing robot adopters	0.002	0.004	0.036
	(0.003)	(0.009)	(0.023)
Never treated	0.002	0.006*	0.025***
	(0.001)	(0.004)	(0.008)

TABLE B4: SUMMARY OF ESTIMATES: ANNUAL AVERAGES

Notes: Table B4 shows the same results as in Table 4 but using as outcomes the average of the last month of each quarter for each year instead of the value for December. The table shows the simple average of estimated $\theta(e)$ event-study coefficients using equation 2 for different periods. Pre-event: years -5 to -1; postevent short run: years 0 to 4; long-run: years 5 to 8. The outcome variable is, in turn, log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panel (b)). The event is defined as a first-time industrial robot importation. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. Refer to notes in Table 4 for the definition of the different samples. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

	Pre-event	Postevent short run	Postevent long run
	(1)	(2)	(3)
A. Log Number formal workers			
Baseline	0.004	0.011	0.051*
	(0.004)	(0.011)	(0.028)
Balanced sample	0.003	0.000	0.045*
	(0.004)	(0.010)	(0.026)
Manufacturing	0.007	0.007	0.078
	(0.009)	(0.018)	(0.049)
Manufacturing robot adopters	0.009	0.010	0.109*
	(0.010)	(0.023)	(0.059)
Never treated	0.012***	0.037***	0.091***
	(0.004)	(0.009)	(0.020)
B. Log Average Wage – formal workers			
Baseline	0.001	0.007*	0.032***
	(0.002)	(0.004)	(0.013)
Balanced sample	0.001	0.007	0.031***
	(0.002)	(0.005)	(0.013)
Manufacturing	0.002	0.000	0.028
	(0.003)	0.008	(0.019)
Manufacturing robot adopters	0.000	0.007	0.047**
	0.003	(0.010)	(0.022)
Never treated	0.001	0.006	0.027***
	(0.001)	(0.004)	(0.009)

TABLE B5: EXCLUDING 2009 AND 2010 COHORTS

Notes: Table B5 shows the same results as in Table 4 but excluding cohorts 2009 and 2010 from the sample. The table shows the simple average of estimated $\theta(e)$ event-study coefficients using equation 2 for different periods. Pre-event: years -5 to -1; postevent short run: years 0 to 4; long run: years 5 to 8. The outcome variable is, in turn, log number of formal workers (Panel (a)) and log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI (Panel (b)). The event is defined as a first-time industrial robot importation. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. Refer to the notes of Table 4 for the definitions of the different samples. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

A: ropulation Censuse	s outcome	es. Long diffe	stences 2010-2	:015								
	All indus	tries										
	Employn	tent-to-pop	Wage-empl	-to-pop	Self-em	ol-to-pop	Labor Foi	ce Particip. rate	Unemplc	yment rate	Share II	nformal
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
MX exposure to robots 2011-2015	-0.2 (0.5)	-0.4 (2.1)	-0.3 (0.5)	-0.5 (1.6)	0.1 (0.4)	0.1 (1.6)	-0.2 (0.5)	-0.3 (1.9)	-0.3 (0.4)	-0.7 (1.2)	0.3 (0.6)	0.4 (1.6)
Observations Adj. R-squared	1,804 0.154	19	$1,804 \\ 0.05$	19	$1,804 \\ 0.180$	19	1,804 0.137	19	1,798 0.026	19	1,797 0.155	19
Mean dep. 2005 I ower limit CI	45.47 -1 3	4 R	23.50 -1 2	-36	21.97 -0.6	-7 1	50.39 -1 2	-4 1	4.406 -0 9	د. 1	49.13 -0 8	с- 8
Upper limit CI	0.9	3.7	0.6	2.6	0.9	2.4	0.7	3.4	0.4	1.7	1.5	3.6
First-stage coefficient First-stage F-statistic	0.721 39.55		0.721 39.55		$0.721 \\ 39.55$		0.721 39.55		0.716 38.75		0.716 38.75	
B: IMSS data - Insured	workers.	Long differe	nces 2010-201	IJ								
	All indus	tries					Manufact	uring industry				
	Ln Numb	er Workers	Ln Pay	roll	Ln Meé	ın Wage	Ln Nur	nber Workers	LnF	ayroll	Ln Mea	n Wage
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)
MX exposure to robots 2011-2015	4.4 (3.6)	5.1 (9.7)	4.7 (4.2)	4.2 (8.5)	0.2 (1.5)	-0.9 (2.7)	-3.4 (6.7)	-25.7 (15.7)	-1.6 (7.4)	-19.2 (15.3)	1.8 (2.7)	6.4 (6.9)
Observations Adj. R-squared Mean den 2005	1,211 0.010 12050	19	1,211 0.014 3 004e+06	19	1,211 0.007 158 1	19	777 0.014 3485	19	777 0.012 954484	19	777 0.011 134 2	19
Lower limit CI	-2.7	-13.9	-3.5	-12.5	-2.7	-6.1	-16.5	-56.6	-16.0	-49.3	-3.5	-7.0
Upper limit CI First-stage coefficient	11.5 0.74	24.2	12.8 0.74	20.9	$3.1 \\ 0.74$	4.3	9.7 0.76	5.2	12.9 0.76	10.9	$7.1 \\ 0.76$	19.9
First-stage F-statistic	38.14		38.14		38.14		36.12		36.12		36.12	

JCI C U E C μ TARTE R6.

	Fo	ormal Worke	ers	Loş	g Average W	age
Model	Pre-event	Postevent	Postevent	Pre-event	Postevent	Postevent
		short run	long run		short run	long run
	(1)	(2)	(3)	(4)	(5)	(6)
CA Only not-yet-treated	0.005	0.016	0.069***	0.001	0.007	0.030**
	(0.004)	(0.012)	(0.029)	(0.002)	(0.004)	(0.013)
SA Last-treated cohort	-0.003	0.038***	0.107***	-0.006	0.014*	0.041*
	(0.013)	(0.013)	(0.036)	(0.007)	(0.007)	(0.023)
CA Never-treated	0.012***	0.041***	0.108***	0.001	0.006	0.026***
	(0.004)	(0.010)	(0.021)	(0.001)	(0.004)	(0.008)
SA Never-treated	-0.027***	0.050***	0.141***	-0.000	0.004	0.022***
	(0.008)	(0.008)	(0.019)	(0.003)	(0.003)	(0.008)
DD Only not-yet-treated	-0.005	0.033*	0.100**	-0.011	0.017	0.049
	(0.015)	(0.019)	(0.046)	(0.011)	(0.013)	(0.032)
DDD Only not-yet-treated	-0.018	0.072*	0.180*	-0.013	0.007	0.020
	(0.038)	(0.039)	(0.096)	(0.014)	(0.017)	(0.041)
DD Never-treated	-0.024***	0.046***	0.134***	-0.003	0.005	0.020***
	(0.010)	(0.010)	(0.021)	(0.004)	(0.004)	(0.008)
DDD Never-treated	0.019	0.018	0.041	-0.006	0.006	0.012
	(0.023)	(0.023)	(0.047)	(0.008)	(0.008)	(0.016)
Synthetic DD		0.071			0.068	
		(0.306)			(0.302)	
Bartik IV		0.051			-0.008	
		(0.097)			(0.026)	

TABLE B7: SUMMARY OF ESTIMATES: DIFFERENT METHODS

Notes: Table B7 presents a summary of results from different models and methods. Sample including cohorts that started importing robots during 2007–2022. Pre-event: years -5 to -1; postevent short run: years 0 to 4; long run: years 5 to 8. The outcome variables are the log number of formal workers and the log average real monthly wage for formal workers, deflated to constant 2010 pesos using Mexican CPI. The event is defined as a first-time industrial robot importation. Clustering of standard errors is at the CZ level. ***, **, and * denote statistical significance at the 1%, 5%, and 10% levels, respectively. *Data sources*: Mexican customs records from Secretaría de Economía. IMMS public data.

	Number of firms	Labor share	Gross value added	GVA per worker	Total fixed assets	Capital per worker	Input cost	Total payroll	Number workers	Workers prod. and	Workers admin	Hours worked	Hours prod. and	Hours admin
			(log)	(log)	(log)		(log)	(log)	(log)	sales (log)	(log)	(log)	sales (log)	(log)
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
Panel (a). All in D_{mt}^{post}	ndustries 0.027* (0.015)	0.019 (0.047)	0.042 (0.028)	0.018 (0.023)	0.001 (0.041)	-0.042 (0.029)	0.052* (0.026)	0.021 (0.029)	0.024 (0.018)	0.022 (0.027)	0.012 (0.034)	0.019 (0.018)	0.013 (0.027)	-0.001 (0.034)
Observations Adj. R-squared Mean	30,397 0.968 62	30,397 0.154 0.476	30,397 0.843 15.92	30,397 0.620 0.0671	30,397 0.803 20.49	30,397 -0.0909 0.0886	30,397 0.874 15.65	30,397 0.854 4.843	30,397 0.905 207.2	30,185 0.860 87.95	20,039 0.763 21.15	30,308 0.901 489.4	30,099 0.851 212.5	19,963 0.751 50.72
Panel (b). Man D_{nt}^{post}	ufacturing 0.056 (0.042)	sector: N -0.014 (0.040)	Metal, ma 0.286* (0.150)	ichinery a 0.119 (0.092)	ind equij 0.324** (0.135)	oment, po 0.015 (0.013)	wer gene 0.242* (0.137)	eration, tr 0.283* (0.143)	ansportatio 0.167* (0.095)	on, plastic 0.225* (0.127)	s and rubb 0.056 (0.130)	er 0.178* (0.102)	0.241* (0.132)	0.093 (0.135)
Observations Adj. R-squared Mean	687 0.972 62.95	687 0.0930 0.480	687 0.776 18.10	687 0.574 0.0561	687 0.744 12.97	687 0.527 0.0795	687 0.778 23.99	687 0.786 8.328	687 0.826 198.9	687 0.789 107.6	469 0.704 22.65	687 0.810 404.1	687 0.770 226	469 0.695 47.61
<i>Notes</i> : Table B8 sho industries and Pank and other manufact is the log number o aggregated to the C assets divided by t bill (column (7)); an production and sale worked in a year. T for the period befor (in thousands), and respectively. <i>Data sc</i>	we the result al (b) for the f uring not inc f firms at the Z-five-digit s he number of d log fixed a: s, and column he sample in e the event, ir shares (ratio, <i>urces</i> : Mexica	s of estima ollowing n luded in Tz CZ-five-d sector level f workers (seets (as a] n (11) for th cludes not n levels, an s). Standau n economi	ting the TV nanufactur able B9. Th, igit sector 1 is labor sha (column (3 proxy for c ne log numl -yet-but-ev id are repoi red errors ar ic censuses	VFE model, ing industri e first depere evel (colum tre, calculate most of work ber of work ventually-tru- tred in milli tre ce dustered 2003, 2008,	equation 3 equation 3 eduator malent varia and (1)). The eduator of the gross valu mm (8)). Co ers in admi eated CZs : at the CZ 2013, and 2	, where the letals, meta ble is the lo following tal wage bil and (c olumn (9) p nistrative jo for 2005–20 for 2005 for	event is de l products, g number o seven depe seven depe l divided b olumn (4)) resents rest obs (includii 19, includii (CP1-adjust re in paren	fined as a f machinery, of firms at tJ andent variation of gross val and GVA 1 ults for the ng managei ng cohorts t theses. ***,	irst-time indu- equipment, e- ae CZ -five-di- bles are balan ue added (col per worker (col per worker (col per worker (col per worker (col per vorker (col)per vorker (col)p	istrial robot i lectronics, a git sector lev ace-sheet me umn (2)); ca olumn (5)); f workers, c (12)–(14) sho nporting rob nporting rob ote statistica	importation. ppliances, pc rel (column (assures captu pital per wor log input co: olumn (10) fc ww similar res oots during 21 f workers, nu 1 significance	Panel (a) prover generation wer generation (1)). The first ring firm pra- ter, calculation sts (column or the log nu- uults for the l 007–2022. M mber of hou s at the 1%, (ssents resu ion, transp dependent oductivity ed as the tr (6)); log to imber of w og number feans are co rs worked 5%, and 10	ts for all ortation, variable and size, tal fixed tal wage orkers in of hours ulculated in a year % levels,

TABLE B8: MECHANISMS: BALANCED SAMPLE

													1	
	Number		Gross	GVA	Total	Capital	,	-	•	Workers		;	Hours	;
	of firms	Labor share	Value Added	per worker	fixed assets	per worker	Input cost	Total payroll	Number workers	prod. and sales	Workers admin	Hours worked	prod. and sales	Hours admin
			(log)	(log)	(log)		(log)	(log)	(log)	(log)	(log)	(log)	(log)	(log)
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)	(6)	(10)	(11)	(12)	(13)	(14)
Panel (a) Man D_{mt}^{post}	ufacturing 0.103** (0.040)	;: Wood a -0.006 (0.027)	nd Paper -0.022 (0.112)	Chemic -0.018 (0.061)	al Indust -0.019 (0.117)	ry 0.032* (0.018)	-0.074 (0.109)	-0.033 (0.114)	-0.004 (0.074)	-0.074 (0.102)	-0.020 (0.142)	-0.015 (0.075)	-0.094 (0.104)	-0.019 (0.146)
Observations R-squared Mean	989 0.974 94.91	989 0.228 0.390	989 0.784 34.43	989 0.611 0.0972	989 0.783 34.37	989 0.604 0.0965	989 0.808 78.99	989 0.758 6.876	989 0.846 299.9	987 0.771 130.6	677 0.648 23.67	989 0.844 726.3	987 0.769 341.2	677 0.627 60.29
Panel (b) Man D_{mt}^{post}	.ufacturinย 0.027 (0.043)	3: Textiles -0.019 (0.044)	s, Appare 0.117 (0.101)	I, Leather 0.056 (0.061)	: Goods , 0.087 (0.118)	Food, and -0.006 (0.007)	l Beverag 0.100 (0.084)	fe Industr 0.146 (0.114)	y 0.061 (0.070)	0.130 (0.103)	0.096 (0.101)	0.074 (0.072)	0.116 (0.107)	0.086 (0.109)
Observations R-squared Mean	1,436 0.942 43.76	1,436 0.0748 0.446	1,436 0.733 17.93	1,436 0.489 0.0456	1,436 0.737 17.59	1,436 0.617 0.0649	1,436 0.772 32.34	1,436 0.740 7.873	1,436 0.808 250.3	1,420 0.754 166.7	736 0.751 29.38	1,436 0.796 525.6	1,420 0.738 360.9	736 0.736 64.90
Panel (c) Non D_{mt}^{post}	manufactu 0.024 (0.014)	rring 0.022 (0.052)	0.035 (0.028)	0.015 (0.023)	-0.009 (0.042)	-0.047 (0.032)	0.050* (0.026)	0.012 (0.027)	0.020 (0.018)	0.015 (0.025)	0.009 (0.035)	0.014 (0.018)	0.007 (0.026)	-0.006 (0.035)
Observations R-squared Mean	27,285 0.968 61.72	27,285 0.162 0.480	27,285 0.850 15.05	27,285 0.624 0.0673	27,285 0.806 20.29	27,285 -0.0817 0.0899	27,285 0.881 12.14	27,285 0.864 4.521	27,285 0.913 201.5	27,091 0.869 81.56	18,157 0.768 20.66	27,196 0.909 480.4	27,005 0.860 199.1	18,081 0.756 49.80
<i>Notes</i> : Table B 9 st- calendar time. Pan following manufac dependent variablı (column (1)). The calculated as the tc total gross value ac (8)). Column (9) pr administrative jobs CZs for 2005–2019, Mexican pesos (CP the CZ level and ar 2018).	turing the rest el (a) presem truring indus e is the log n following se tral wage bill lded (columr esents result s (including r including c including c including c e in parenthe	ults of estimults of estimults for the stresults for the stress textilution tumber of fi further of fi further of fi further of the log s for the log managers). The stress set of the set	r the follow e, clothing, e, clothing, rms at the ' thent variab dent variab va	TWFE mod- ving manufi, leather, wo CZ-five-dig les are bala. e added (co cker (column f workers, o f workers, o 12)–(14) sho oorting robo number of v te statistical	el, equatio acturing in od, paper, jrit sector le nce-sheet 1 nce-sheet 1 n(5)); log i olumn (2)); w similar the during ' workers, th workers, th	n 3, where 1 dustries: foo printing, an vel (columr measures ca mput costs (1 for the log results for H 2007–2022.] ne number c ce at the 1%	the event is od, drinks, d nonmeta n (1)). The i pturing fir worker, cal worker, cal worker, cal number of he log num Means are of hours wo , 5%, and 1	s defined as tobacco, en ullic mineral first depend m producti culated as t vorkers in workers in ber of hour calculated f orked in a y 0% levels, r	a first-time i ergy, fossil fu products. Pa lent variable. vity and size, he total fixed age bill (colu production a s worked in a s worked in a ear (in thous. espectively. L	ndustrial ro els, chemica nel (c) inclu as the log nu aggregated assets divid mn (7); and nd sales, anc ry ear. The s before the e ands), and s anta sources:	bot importati ls, plastics, ru des all nonme mber of firms to the CZ-fiv ed by the num log fixed asse l column (11) ample includd vent, in level. hares (ratios). Mexican Econ	on, and bala bber. Panel unufacturing a at the CZ-1 re-digit sect ther of work for the log n es not-yet-bu s, and are re Standard e Standard e	ancing the (b) for resu five-digit s sudurties or level: la kers (colurr vy for capit uurber of v uurber of v uur-eventual ported in r rrors are cl ses (2003, 2	the first for the first for the first for the first ector level bor share, in (3)); log up, column vorkers in vorkers in vorkers in vorkers in ustered at ustered at ustered at ustered at 008, 2013,